Check for updates

OPEN ACCESS

EDITED BY Jinwei Zhang, University of Exeter, United Kingdom

REVIEWED BY

Andrew K.J. Boyce, University of Calgary, Canada Bidisha Sinha, Indian Institute of Science Education and Research Kolkata, India Luis Eduardo M. Quintas, Federal University of Rio de Janeiro, Brazil

*CORRESPONDENCE George D. Bittner, ⊠ bittner@austin.utexas.edu

SPECIALTY SECTION This article was submitted to Membrane Physiology and Membrane Biophysics, a section of the journal Frontiers in Physiology

RECEIVED 09 December 2022 ACCEPTED 14 February 2023 PUBLISHED 15 March 2023

CITATION

Mencel ML and Bittner GD (2023), Repair of traumatic lesions to the plasmalemma of neurons and other cells: Commonalities, conflicts, and controversies. *Front. Physiol.* 14:1114779. doi: 10.3389/fphys.2023.1114779

COPYRIGHT

© 2023 Mencel and Bittner. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Repair of traumatic lesions to the plasmalemma of neurons and other cells: Commonalities, conflicts, and controversies

Marshal L. Mencel¹ and George D. Bittner^{2*}

¹Institute of Cell and Molecular Biology, University of Texas at Austin, Austin, TX, United States, ²Department of Neuroscience, University of Texas at Austin, Austin, TX, United States

Neuroscientists and Cell Biologists have known for many decades that eukaryotic cells, including neurons, are surrounded by a plasmalemma/ axolemma consisting of a phospholipid bilayer that regulates transmembrane diffusion of ions (including calcium) and other substances. Cells often incur plasmalemmal damage via traumatic injury and various diseases. If the damaged plasmalemma is not rapidly repaired within minutes, activation of apoptotic pathways by calcium influx often results in cell death. We review publications reporting what is less-well known (and not yet covered in neuroscience or cell biology textbooks): that calcium influx at the lesion sites ranging from small nm-sized holes to complete axonal transection activates parallel biochemical pathways that induce vesicles/membranebound structures to migrate and interact to restore original barrier properties and eventual reestablishment of the plasmalemma. We assess the reliability of, and problems with, various measures (e.g., membrane voltage, input resistance, current flow, tracer dyes, confocal microscopy, transmission and scanning electron microscopy) used individually and in combination to assess plasmalemmal sealing in various cell types (e.g., invertebrate giant axons, oocytes, hippocampal and other mammalian neurons). We identify controversies such as plug versus patch hypotheses that attempt to account for currently available data on the subcellular mechanisms of plasmalemmal repair/sealing. We describe current research gaps and potential future developments, such as much more extensive correlations of biochemical/biophysical measures with sub-cellular micromorphology. We compare and contrast naturally occurring sealing with recently-discovered artificially-induced plasmalemmal sealing by polyethylene glycol (PEG) that bypasses all natural pathways for membrane repair. We assess other recent developments such as adaptive membrane responses in neighboring cells following injury to an adjacent cell. Finally, we speculate how a better understanding of the mechanisms involved in natural and artificial plasmalemmal sealing is needed to develop better clinical treatments for muscular dystrophies, stroke and other ischemic conditions, and various cancers.

KEYWORDS

plasmalemmal/axolemmal disruptions, axonal severance, apoptosis, membrane repair, polyethylene glycol, calcium

1 Introduction to plasmalemmal sealing

1.1 A brief history of plasma membrane repair

Eukaryotic cells in multicellular organisms often incur damage to their plasmalemma, the membranous diffusion barrier between the extracellular and intracellular fluids that completely surrounds each individual cell (Singer and Nicolson, 1972; Leonard, 1993; Alberts et al., 2015). The plasmalemma typically consists of a phospholipid bilayer studded with proteins, as do other membrane-bound organelles. Water molecules bound to the inner and outer surface of hydrophilic phospholipids and proteins of the plasmalemma and other membranous vesicles/ organelles prevent their spontaneous fusion. For example, membrane-bound water prevents the plasmalemma of closely apposed cells in a tissue from spontaneously fusing, thereby preventing tissues composed of individual cells from forming a multinucleate cytoplasmic syncytium with no plasmalemmal boundaries. Membrane-bound water also prevents intracellular vesicles and organelles from readily fusing with each other and/ or the plasmalemma (Singer and Nicolson, 1972; Alberts et al., 2015).

Traumatic damage to a plasmalemma ranges from a small hole of nanometer-to micron-sized diameters to complete transections of a cell body or its cytoplasmic extensions (i.e., filopodia, dendrites, axons, etc.) that may be many (2–1,000) microns in diameter (LaPlaca, 2009; Vargas and Bittner, 2019; Kang et al., 2020). Prior to 1994, if covered at all in textbooks, repair of a complete transection of an intact cytoplasmic extension (Figure 1A) was assumed to occur by collapse and fusion of the



FIGURE 1

Topographical models of previously assumed mechanisms of axonal sealing. (A) An axon completely transected on the far right prior to any repair processes. (B) Sealing of a transected axon by collapse and fusion of apposed plasmalemmal leaflets. (C) Sealing of a transected axon by spontaneous formation of a membranous partition-like structure. (D) Sealing of a very small (nm) membrane disruption by the thermodynamically favorable lateral migration of lipids. apposing plasmalemmal leaflets at each cut end (Figure 1B) or the formation of a membranous partition (Figure 1C) (Yawo and Kuno, 1983; Yawo and Kuno, 1985; Kelly, 1985, 1991). A smaller hole anywhere in the plasmalemma was assumed to repair by the spread and fusion of lipid bilayers surrounding the hole (Figure 1D) (reviewed in McNeil and Baker, 2001; Horn and Jaiswal, 2019). Such repairs were assumed to take microseconds to milliseconds to completely seal the damaged plasmalemma, i.e., to restore diffusion properties of an intact plasmalemmal barrier, especially to prevent the influx of extracellular calcium (Schlaepfer and Bunge, 1973; Krause et al., 1994; reviewed by Vargas and Bittner, 2019).

In 1994, two laboratories (Krause et al., 1994; Steinhart et al., 1994), working without the knowledge of the other's research efforts, independently reported that cells seal holes in, or complete transections of, a eukaryotic plasmalemma in oocytes, axons, or epithelial cells by an accumulation, migration, and fusion of vesicles and other membranous structures (Figure 2). These and subsequent studies by various laboratories (reviewed in part by Moe et al., 2015; Vargas and Bittner, 2019) show that many (perhaps all) eukaryotic cells exhibit an influx of extracellular calcium (Ca²⁺) at the wound site that is required to initiate sealing. This influx of extracellular Ca^{2+} activates parallel biochemical pathways, intracellular organelles common to vesicular trafficking in the Golgi apparatus, conventional exocytosis, and coordinated exocytosis at chemicalreleasing synapses (Figures 2, 3).

These repair processes shared by all eukaryotic cells (partially reviewed by Vargas and Bittner, 2019) have some conflicting details as reported for different cell types (e.g., oocytes vs. neurons vs. epithelial cells) and several controversies on specific sealing mechanisms (e.g., plug vs patch, vesicle shedding vs. lipid spreading). That is, sealing dynamics by vesicular interactions that occur after onset of plasmalemmal trauma take seconds to minutes in different cell types to restore a plasmalemmal seal having diffusion barrier properties observed for their originally-intact plasmalemma. These vesicular interactions could form a vesicular "plug" that produces a constantly changing diffusion barrier that gradually attains the barrier properties of the intact plasmalemma (Figures 2A-C, E, F; Bittner et al., 2016b). Alternatively, vesicles could interact to form a large wound vesicle that immediately seals the membrane as a "patch" in one final step to attain the diffusion properties of an intact membrane (Figures 2A, B, D, E, G; Andrews et al., 2014; Davenport et al., 2016). Furthermore, several studies have also reported that amphoteric substances like polyethylene glycol (PEG) can restore a plasmalemmal barrier exceedingly rapidly, perhaps within microseconds to milliseconds via a mechanism similar to original hypotheses for sealing plasmalemmal transections by direct fusion of plasmalemmal leaflets (Figure 1B) or holes by lipid spreading (Figure 1D) (Spaeth et al., 2012b; Zuzek et al., 2012; Vargas and Bittner, 2019).

1.2 Structure of this review

In this review, we summarize data on initial, intermediate, and final steps of plasmalemmal sealing to begin to correlate cellular/molecular with biochemical/biophysical events that are



Models of natural plasmalemmal repair mechanisms. (A) Cellular components and events involved in vesicle-mediated membrane repair. Influx of Ca²⁺ through membrane disruptions leads to the formation of membranous structures that aggregate at the site of injury. The Golgi apparatus, smooth endoplasmic reticulum (SER) and mitochondria can all contribute to vesicle formation and subsequent repair processes. Vesicles can arise from any membranous source including the plasmalemma, organelles, and surrounding glia/myelin. Vesicles continue to aggregate at the site of injury (B) and form functional barrier either through a plug (C) or patch (D) mechanism. Regardless of mechanism, repaired cells eventually form a continuous bilayer at the site of previous injury (E). Schematic representation of the events described by the plug (F) and patch (G) hypothetical mechanisms. The plug hypothesis features progressive formation of a diffusion barrier by dense packing of vesicles followed by formation of a continuous plasmalemma. The patch hypothesis involves formation of a large wound vesicle that seals in a single-step fusion event followed by the collapse of this structure (explodosis). The outer plasmalemma is shown in gray while intracellular membrane-bound components (i.e., vesicles, organelles, etc.) and endocytotic vesicles are shown in orange. Adapted from Vargas and Bittner (2019).

common to sealing in all eukaryotic cells. We also identify data that describe conflicting specific properties of different eukaryotic cell types and describe controversies regarding how current data on plasmalemmal sealing should be interpreted (Summarized in Table 1).

This review first assesses the general appropriateness, specific applications, and controversies of different techniques and measures used to assess the state of a plasmalemmal seal (Section 2), followed by a brief description of the model systems primarily used to study sealing after traumatic membrane damage (Section 3). Section 4 examines hypotheses and data (or the lack thereof) prior to 1994 for sealing of plasmalemmal holes by direct lipid spread or fusion of plasmalemma leaflets following complete transection of cytoplasmic extensions. Section 5 describes the data for the role of Ca2+ and vesicles in plasmalemmal sealing and some unanswered questions such as whether vesicles are moved by bulk flow and/or molecular motors. Section 6 describes various biochemical processes and substances involved in vesicle-mediated natural sealing and emphasize that most are not yet correlated with subcellular events in membrane repair. Section 7 describes and assesses data that do, or do not, support the validity of the plug vs. patch hypotheses and how neither hypothesis for this intermediate

event in cell sealing yet accounts for the ultrastructural morphology or biophysical properties of the final state of membrane repair. Section 8 describes how a previous traumatic lesion may facilitate the sealing of a subsequent lesion. Section 9 describes how PEG acts as a direct membrane fusogen and may repair plasmalemmal damage by mechanisms identical to those originally hypothesized prior to 1994. In Section 10, we present our assessment of the current state of the field and suggest experiments that might provide data needed to resolve some current controversies.

2 Measures of plasmalemmal sealing and their limitations

2.1 Electrophysiological

Membrane potential (V_m) and input resistance (R_{in}) are two measures used to assess the state of a membrane seal in a variety of cell types, although these measures have not been used in most recent studies (see Table 1). These electrophysiological measures typically assume that a restoration of V_m and R_{in} to levels of uninjured membranes represent 100% sealing efficacy and lesser values are linearly related to the effectiveness of the seal as a diffusion barrier (Krause et al., 1995; Bittner and Fishman, 2000).

These electrophysiological (V_m, R_{in}) measures of the status of a membrane seal may be reasonably accurate for cells where all plasmalemmal extensions are less than a space constant from the intracellular recording and stimulating electrode(s). However, many cells (or long cytoplasmic extensions such as axons or dendrites) often have membrane cable properties distributed over many space constants. For these cells or cytoplasmic extensions, measurements of V_m and R_{in} are often very insensitive to changes in R_{in} at the lesion site, even when measurements are taken within 0.01 of a space constant from the transection or lesion site (Krause et al., 1995). That is, values obtained for R_{in} are often not reliable measures of seal effectiveness (Bittner and Fishman, 2000).

In contrast to measures of Rin, measures of input current density (Ii) do not depend upon axon cable properties. Hence, the extracellular measurement of Ii by a vibrating probe combined with intracellular measures of V_m provide a more reliable measure of the effectiveness of an axonal seal (Krause et al., 1995). Formation of a seal equivalent to that of an intact membrane is considered to have occurred when I_i declines to baseline values and V_m recovers to values obtained from the uninjured (control) axon. Measures of Ii must be paired with measures of V_m that show that V_m is equal to control values for the intact plasmalemma before the cell was injured (Bittner and Fishman, 2000). That is, I_i shows a marked decline as V_m decreases in an electrically dysfunctional axon with a poorly sealed plasmalemmal lesion. In short, both a recovery of V_m combined with a decline of Ii to values measured for the intact plasmalemma strongly suggest restoration of barrier permeability to that of the originally intact plasmalemma. Such comprehensive assessments have been made only for completely transected invertebrate giant axons (Krause et al., 1994; Fishman et al., 1995; Eddleman et al., 1997; Eddleman et al., 1998a; Eddleman et al., 2000).

2.2 Fluorescent tracers

The extracellular and/or intracellular addition of fluorescent labels have been used to assess the plasmalemma sealing of most cell types reviewed herein. Localization and labeling of dyes are typically observed *via* fluorescence and/or confocal microscopy. The formation of a tracer dye barrier usually correlates well with the formation of a seal as assessed by R_m , V_m and I_i —and tracer data are easier to obtain than I_i (Bittner and Fishman, 2000). However, dye tracers almost always serve as more qualitative than quantitative measures, especially when used to assess the state of a seal in an individually wounded cell (see Section 3.4). When using fluorescent tracers, several variables must be considered to accurately assess the state of a plasmalemmal seal.

2.2.1 Tracer targets: Hydrophilic vs. hydrophobic vs. Ca²⁺

Hydrophilic dye added to the solution bathing an axon is excluded from successfully sealed cells (Eddleman et al., 1997). Alternatively, hydrophilic dye can be loaded into cells by microinjection. If cell-loading is followed by hydrophilic dye addition to the extracellular solution, a clear boundary can be seen between the contrasting dyes (Spötl et al., 1995; Eddleman et al., 1997; Bittner and Fishman, 2000; Chuo et al., 2018). Cells can also be loaded with calcein-AM that is transported into the cell and fluoresces intracellularly after cleavage of the AM moity (Eddleman et al., 1995). Styryl (lipophilic) dye added to the bath prior to wounding is readily incorporated into the plasmalemma and/or myelin (Betz et al., 1992). Labeling the membrane enables observation of endo- and exocytotic events contributing to formation of a seal (Chuo et al., 2018). Fluorescent tracers are also used to monitor ionic dynamics in cells, such as Ca2+-sensitive Fura-2 (Steinhardt et al., 1994; Togo, 2019; Sridhar et al., 2020). In oocytes and mammalian cell culture, GFP fusion proteins have been used to study the biochemical agents involved in sealing (Vaughan et al., 2014).

2.2.2 Molecular weight of tracers affect diffusion across plasmalemmal lesions

Barrier formation to exclude the membrane dynamics is associated with sealing (Benink and Bement, 2005; Corrotte et al., 2013). Influx/efflux of various markers depends on the molecular size of the dye molecule used (Gallant and Galbraith, 1997; Eddleman et al., 2000). For example, crayfish MGAs initially exclude Texas Red-dextran (70 kDa, then 3 kDa) followed at least 30 min later by Calcein (0.6 kDa) when the two are added to the extracellular bath (Eddleman et al., 2000; See Section 3.2). Exclusion of ions take yet longer (Krause et al., 1994). Thus, in the cell types in which it has been studied, there is a progressive exclusion to molecules of higher to lower molecular weight as a barrier (seal) forms (reviewed in Vargas and Bittner, 2019).

2.2.3 Electrical charge of tracers affect diffusion across plasmalemmal lesions

Properties of the intracellular environment can affect the uptake of fluorescent tracers. For example, unsealed crayfish MGAs exclude negatively-charged, but not positively-charged, dye molecules (Eddleman et al., 1998b). An outward axoplasmic flow in transected squid GAs can exclude both neutral and negatively charged dye molecules (Eddleman et al., 1998a; Eddleman et al., 2000).

2.3 Morphological observation

Morphological observations alone almost always yield ambiguous measures of plasmalemmal sealing, in large part because they usually do not have the resolution to image the 4-10 nm wide membrane bilayer. In general, high magnification (>80,000X) transmission electron microscopy (TEM) is needed to detect whether a membrane is continuous and intact. Even then, it is almost impossible to collect and analyze many serial sections and develop a 3D reconstruction of the membrane topology (Vargas and Bittner, 2019). To the best of our knowledge, this technique has never been utilized to show a continuous membrane as the result of plasmalemmal disruptions more than 1 μ m in diameter. Studies claiming *de novo* formation of an intact membrane at the site of injury (See Section 4) using 5000X TEM (Spira et al., 1993) or 100X phasecontrast imaging (Yawo and Kuno, 1985) do not have the resolution to justify a conclusion of a fully formed diffusion barrier.

Scanning electron microscopy (SEM) has also been used to observe the accumulation of vesicles at the transected ends of axons (Eddleman et al., 2002) and lesion sites in sea urchin eggs (McNeil and Baker, 2001) as discussed in detail in Section 4. SEM observations do not, however, provide a reliable measure of the formation of a complete seal due to their inability to monitor in real time paired with measures of dye exclusion or V_m , R_{in} , and I_{in} . Additionally, vesicles may collapse or explode as a result of the dehydration and other processes applied during TEM or SEM sample preparation (Chuo et al., 2018; Vargas and Bittner, 2019).

In summary, the most reliable assessments of plasmalemmal sealing use a combination of the measures described above, including serial confocal measures of fluorescent dye exclusion, V_m and I_{in} , and high-resolution TEM images taken through the lesion site. These multiple-method measures of membrane repair have primarily been studied using invertebrate giant axons (GAs), namely, squid, crayfish, and earthworm GAs (Krause et al., 1994; Ballinger et al., 1997; Eddleman et al., 1997; Godell et al., 1997; Eddleman et al., 2000; Lichstein et al., 2000; Bittner et al., 2016c; Vargas and Bittner, 2019).

3 Use of model systems for studying plasmalemmal repair

3.1 Individual echinoderm and amphibian eggs

Unfertilized oocytes of sea urchins (Steinhardt et al., 1994), starfish (Fein and Terasaki, 2005), Xenopus (Mandato and Bement, 2001; Clark et al., 2009), and mice (Um et al., 2020), of 50-1000 µm in diameter are often used as model systems to assess sealing of plasmalemmal lesions of less than 1 µm to greater than 5 µm in diameter. Vesicles and other sub-cellular organelles in these cells are easily visualized with fluorescence or confocal microscopy and the cells penetrated with microelectrodes. These large unfertilized oocytes have many intracellular vesicles docked just interior to the plasma membrane and can rapidly interact with each other and the plasmalemma to seal a plasmalemmal lesion within seconds to minutes after natural (sperm) or artificial trauma producing a plasmalemmal hole microns in diameter. Due to such rapid repair, electrophysiological measurements and fluorescent analyses of unfertilized oocytes cannot easily or accurately assess the transition from injury to sealing. Hence, rapid sealing measurements from such unfertilized eggs may lead to inappropriate interpretations of an immediate (step-wise change) barrier formation. The egg models, fertilized or unfertilized, are also often regarded by cell biologists and others as generic representations of eukaryotic cells, compared to axons (described below) that are often regarded as specialized cells not generally representing eukaryotic cells. This assumption may not be appropriate for studies of plasmalemmal repair due to the highly specialized wound response observed in oocytes. That is, almost all other cell types do not have an abundance of docked cortical vesicles and seal within minutes to hours (reviewed in Hendricks and Shi, 2014; Vargas and Bittner, 2019).

3.2 Individual invertebrate giant axons

Cytoplasmic extensions (axons, neurites) of invertebrate nerve cells are often used as a model system to assess sealing of small holes or complete transections of membrane-bound structures. These cytoplasmic extensions do not have an abundance of vesicles docked at the membrane (Bittner et al., 2016b). Invertebrate giant axons (GAs) are large enough (50-1,000 µm) to make intra- and extra-cellular electrophysiological recordings from single cells and easily visualize subcellular organelles via confocal fluorescence. Additionally, transection of GAs avoids damage to the cell body and its nucleus. Invertebrate GAs have a less rapid sealing response (minutes to hours) to traumatic lesions than unfertilized eggs (seconds to minutes; see Table 1). Hence, the progression of repair processes in invertebrate GAs can be observed with better temporal and spatial resolution. For example, different molecular sized tracers (discussed in Section 2.2.2) are excluded at differing times post transection, a result which would be difficult to impossible measure in rapidly sealing unfertilized eggs (Eddleman et al., 2000). The large size of invertebrate GAs also enables control of some known factors that affect plasmalemmal sealing, such as induction of Ca2+-induced vesiculation via internal microdialysis of Ca²⁺ (Fishman and Metuzals, 1993). Data from such invertebrate GA models consistently show progressive re-formation of a diffusion barrier similar to an intact plasmalemma, in contrast to the near-instantaneous formation often inferred from oocyte studies (Abe and Cavalli, 2008; Spaeth et al., 2010; Bittner et al., 2016b).

3.3 Individual mammalian neurites in culture

Cultured mammalian PC12 or B104 cells as a model system do not exhibit instantaneous sealing and show progressive barrier formation (Yoo et al., 2003), as may most eukaryotic cell types. These cells can be easily individually identified and 20-40 observed in their entirety (i.e., including the nucleus and cytoplasmic extensions) at the same Petri dish. This property can facilitate relatively high throughput comparisons of cell repair under various wounding and environmental conditions. For example, B104 neurites transected closer to the soma successfully seal at lower rates than those transected further away from the cell body (Yoo et al., 2004; Nguyen et al., 2005; McGill et al., 2016). However, these cells are too small in diameter $0.5-2 \,\mu\text{m}$) to easily observe cytoplasmic organelles or penetrate with microelectrodes. These cells are typically utilized to assess data on the biochemical pathways that produce a plasmalemmal seal following neurite transection (Detrait et al., 2000; Spaeth et al., 2012a; Spaeth et al., 2012c; Spaeth et al., 2012b; Zuzek et al., 2012; Poon et al., 2018).

3.4 Populations of damaged cells in culture

The use of damaged cells in culture to assess plasmalemmal sealing after traumatic injury involves shearing/scraping of single layer cell cultures (often mammalian epithelial cells) or syringe

| | | Uocytes | Mammalian neurites | Mammalian epitnelial cells | Muscie celis |
|--|--|---------------------------------|---|--|---|
| Electrophys-iological R _{in} , V _m , I _i) | 5,6,7,9,10 | | 19,20,23 | | |
| gluorescence | 5,6,7,8,10 | 16,17,18 | 19,24,27,28,30 | $31, 32, 35^{\rm H}, 40, 4^{\rm H}, 44^{\rm H}, 47, 49, 51$ | 55,59,60,61 ^H ,65 |
| Confocal | 1,3,4,5,6,9,11 | 14,17,18 | 20,22,23,29 | $31,33,35^{H},36,37,39,43^{H},45,46,49,50^{H},51,52,53$ | 55,56,58,65 ^H ,66 ^H |
| mmunostaining | 3 | 14,16 | 22 | $37,40,43^{\rm H},44^{\rm H},47$ | $55,56,58,61^{\rm H},63^{\rm H},64^{\rm H},65,66^{\rm H}$ |
| SEM | 2 | 15 | | 33 | |
| TEM | 1,4,6,8,9,10,11 | 17 | 20 | $32,33,35,37,43^{\rm H},46,49$ | 57 ^H ,58,66 ^H |
| ble 1. Methods of assessing the stat | te of a membrane seal. Numbers in the ta | uble represent original researc | ch articles that have used these methods to ass | sess membrane repair in the various model systems (see Suppl <mark>en</mark> | nentary Materials for corresponding list o |

references). Note that several model systems have not been assessed by all available methods (especially by electrophysiological or scanning electron microscopy (SEM)) or have been assessed using certain methods in only one publication. A superscript H (X^t) indicates the referenced paper used human cells as a model system that

10.3389/fphys.2023.1114779

uptake of sea urchin eggs. In the former procedure, cells are grown to ~70–90% confluency and the culture dish is scraped to induce widespread, unquantified plasma membrane damage to many cells (Miyake and McNeil, 1995; Terasaki et al., 1997; McNeil and Steinhardt., 2003; Mellgren et al., 2007). In the latter procedure, many sea urchin eggs are taken up into a syringe and subsequently expelled to yield unspecific and unquantified damage to many cells (McNeil and Baker, 2001). The problem in using these models to assess the averaged sealing response is a variety of lesion types across a population of cells of unknown number, likely containing a significant number of cells wounded once or many times and cells without a plasmalemmal disruption.

4 Previously-assumed mechanisms of plasmalemmal sealing: Plasmalemmal collapse and lipid spreading

Prior to 1994, the subcellular, biochemical, etc., mechanisms that repair a damaged plasmalemma had not been directly studied. Lipid spreading was often assumed to repair small holes and collapse and fusion of apposed plasmalemmal leaflets were assumed to rapidly seal a complete transection resulting in a continuous plasmalemma whose chemical composition would be similar to the intact membrane (Schlaepfer and Bunge, 1973; Krause and Bittner, 1990; Krause et al., 1994; Vargas and Bittner, 2019).

The sealing of micro-punctures in the plasma membrane were assumed to be repaired by lipid spreading (Kelly, 1985). In this model, the fluid-like mobility of membrane lipids would allow their lateral migration to close the disruption in a thermodynamically favorable process (Figure 1D). However, lateral lipid mobility is now known to be greatly restricted in biological membranes (Golan et al., 1984; Gordon et al., 1985; Kwik et al., 2003) and lipid spreading has only been convincingly demonstrated in artificial unilamellar vesicle membranes (Riske and Dimova, 2005). Nevertheless, lipid spreading was occasionally mentioned as a potential mechanism of plasmalemmal repair (McNeil and Baker, 2001; McNeil and Terasaki, 2001), primarily as a proposal to seal very small plasmalemmal disruptions on nanometer scales (McNeil and Baker, 2001; Horn and Jaiswal, 2019).

Larger-scale plasmalemmal damage of transected cytoplasmic processes was originally assumed to seal by collapse and subsequent fusion of cut plasmalemmal leaflets (Figure 1B) (Kelly, 1985), although no studies were referenced to support the assumption. A few studies on invertebrate axons [squid GAs (Gallant, 1988), cockroach GAs (Meiri et al., 1981), and Aplysia (Spira et al., 1993)] and mammalian axons [rat septal neurons (Xiea and Barrett, 1991) and spinal axons (Kao et al., 1983)] interpreted their data according to this collapse and fusion hypothesis. However, the data in these studies were obtained using techniques that lacked the necessary resolution to support their interpretation (Bittner and Fishman, 2000; Spaeth et al., 2010; Vargas and Bittner, 2019).

One alternative model prior to 1994 arose from studies on the cockroach GA (Yawo and Kuno, 1983; Yawo and Kuno, 1985). This model was based on observations that V_m and I_i recovered to values representative of a successful membrane seal. Accompanying low-resolution phase contrast microscopy was used to claim the

TABLE 1

Methods of assessing membrane repair.

TABLE 2 Results of sealing experiments.

| | Invertebrate GAs | Oocytes | Mammalian neurites | Mammalian epithelial cells | Muscle cells |
|------------------------------------|---|---|---|---|---|
| Sealing time | Squid GA: ~50min ^{a(4,9,10,12)} | Xenopus oocyte: <30–60s ^(14,18) | Rat astrocytes: ~15min ⁽¹⁹⁾ | HeLa: ~300-400s ^(44H) | C2C12 Myoblasts: 100-200s ^(55,56) |
| | Earthworm GA: ~60min ^(1,10,11) | Unfertilized sea urchin egg: 15-30s ^(15,16,17) | B104 cells: 2-6min ^(26,28,29,30) | MDCK: ~50s ^(52,53) | Mouse myofibers: ~160s ⁽⁵⁹⁾ |
| | Crayfish MGA: ~60min ^(4,6,9) | | PC12 cells: 5-10 min ⁽²⁹⁾ | 3T3 Fibroblasts: ~90s ⁽⁵¹⁾ | |
| Ca ²⁺ requirement | 1,2,3,4,5,7,9,10,12 | 14,15,16,17,18 | 20,23,24,25,26,28,29,30 | 35,37,43 ^H ,47,50 ^H ,52,53,54 | 55,58,59 |
| Annexins | | 14 | 20,22 | 31,32,35,38,39,4 ^H | 57 ^H ,59,62,66 ^H |
| Oxidation pathway | 9 | 14 | 19,24,25,27 | 42,48,49 | 55,56,57 ^H ,60,61 ^H ,62,63 ^H ,64 ^H ,66 ^H |
| Phospholipase pathway | 3,9 | 14,18 | 24,30 | 46,51 | |
| Calpains | 9 | | 29 | 42,47 | 61 ^H ,63 ^H |
| cAMP | 3,9 | 14,18 | 19,21,24,27,30 | 33,46,51,52 | |
| ESCRT | | | | 34 ^H ,44 | 65 |
| Synaptic fusion proteins | 3 | 16 | 22,23,26,27,29,30 | | 64 ^H |
| Changes in local lipid composition | | 13,14,18 | 30 | 45,48,50 ^H | 55,59 |
| Cytoskeletal dynamics | 9 | 16,18 | 24 | 31,33,36,37,41 ^H ,43 ^H ,47,54 | 55,58,60 |

Sealing times and biochemical pathways found to be involved in the sealing process across model systems. Numbers in the table represent original research articles publishing these results (see Supplemental Materials one for the full list). A superscript H (X^H) indicates that the referenced paper used human cells as a model system.

^aComplete seal formation has only been observed in squid GAs treated with exogenous calpain

establishment of a high resistance, partition-like bi-layered membrane proximal to the transected end (Figure 1C). This morphometric protocol lacked the appropriate resolution to identify the structure as a lipid bilayer.

5 Currently-documented mechanisms of plasmalemma sealing

5.1 Ca²⁺-mediated vesicle interactions

The rapid formation of membranous structures has long been reported to occur in response to axonal transection (Ramón Y Cajal et al., 1991). Injury-induced vesicles had been observed to occur after traumatic injury to squid GAs (Fishman et al., 1990). The role of such vesicles was unknown until 1994. In 1994, Krause et al. (1994) reported that transected squid and earthworm GAs seal by an accumulation of vesicles induced by Ca2+ influx. That same year, Steinhardt et al. (1994) identified Ca²⁺-induced vesicle accumulation as the primary sealing mechanism in mechanically-wounded unfertilized sea urchin eggs and mammalian fibroblasts. These papers agreed that plasmalemmal repair primarily required influx of Ca²⁺ through a membrane disruption (Krause et al., 1994; Ballinger et al., 1997; Eddleman et al., 1997). Intracellular Ca²⁺ can also increase by influx through voltage-gated ion channels in the plasma membrane (George et al., 1995; Sattler et al., 1996) or release from intracellular stores (See Section 6.3). However, these mechanisms probably secondarily contribute to plasmalemmal sealing/repair, with influx through the lesion site remaining the primary cause of increased intracellular Ca²⁺ (Strautman et al., 1990; Fishman et al., 1995; Ziv and Spira, 1995; Bittner G. D. et al., 2000).

Microdialysis of squid GAs showed that high cytoplasmic concentrations of Ca2+ produced vesiculation (Fishman and Metuzals, 1993). Various cellular origins of these Ca²⁺-induced vesicles have been proposed and cells most-likely utilize any membranous structures readily available for plasmalemmal repair (Bittner G. et al., 2000; Bittner and Fishman, 2000). For example, in unmyelinated axons, vesiculation occurs as a result of Ca²⁺ influx from complete transections or mechanically induced lesions that induce membranous structures primarily by endocytosis of the axolemma (Ballinger et al., 1997; Eddleman et al., 1997; Eddleman et al., 1998a). Muscle fibers exhibit formation of dysferlin-containing vesicles derived from the sarcolemma (McDade, Archambeau, and Michele, 2014). Kidney epithelial cells repair plasmalemmal lesions by vesicles formed by a breakup of the SER (Reddy et al., 2001). Transected earthworm GAs often form multilayered vesicles that appear to arise by delaminations of their myelin sheath (Ballinger et al., 1997). Additionally, Ca2+ entry causes mitochondria, lysosomes, and preexisting vesicles to migrate towards the injury site (Rintoul and Reynolds, 2010). In more specialized cells, such as oocytes, pre-formed cortical vesicles facilitate repair processes by rapid migration to the lesion site (Terasaki et al., 1997; Mandato and Bement, 2001).

These data summarized above strongly suggest that eukaryotic cells derive vesicles from any and all readily available membranous structures to facilitate wound repair in an evolutionarily conserved mechanism (Eddleman et al., 1997; Spaeth et al., 2010) While the

specific final cellular mechanism involved in vesicle-mediated repair (plug vs. patch: see Section 7) is unknown and debated, the biochemical pathways involved (See following section) appear common to sealing of holes and transections in all the model systems reported in Section 4 and mechanisms reported in this Section.

5.2 Cytoskeletal reorganization and function

The cortical cytoskeleton has many functions in response to, and repair of, plasmalemmal damage-most prominently aiding in the delivery of vesicles to the lesion site and in the stabilization of the membrane bordering the injury. Plasmalemmal injury disrupts the local cytoskeleton at the wound site and triggers disorder of adjacent structures, perhaps involving active processes of disorganization. For example, microtubule (MT) disassembly is correlated with Ca²⁺ influx at the wound site that directly promotes the detachment of MTs from the plasmalemma (Togo, 2006). Following this disassembly, MTs elongate toward the site of injury. Similarly, Ca²⁺ influx at the wound site drives remodeling of intracellular actin networks by depolymerizing actin filaments to G-actin (Bíró and Venyaminov, 1979; Li et al., 2018; Ebstrup et al., 2021). This reorganization of the cytoskeleton assists the transport kinesinmediated delivery of vesicles and lysosomes along MTs to the injured portion of plasmalemma (Cabukusta and Neefjes, 2018). MT-independent transport along actin filaments may also occur in long-distance delivery of membranous structures (Schuh, 2011; Ebstrup et al., 2021). For example, actin filaments in muscle cells help deliver membranous structures to sites of sarcolemmal injury (McDade, Archambeau, and Michele, 2014).

In addition to its role in vesicle transport, cytoskeletal reorganization also helps mediate vesicle interactions at the plasmalemma. The cortical actomyosin network facilitates both neuronal and non-neuronal exocytosis (Miklavc and Frick, 2020). Nanoscale remodeling of the cortical actin network aids in the anchoring, priming, and fusion of vesicles with the plasmalemma. Actin networks assist the organization of SNARE proteins and Ca²⁺ channels that comprise the cellular secretory machinery. For example, localization of the synaptic fusion protein SNAP-25 (see Section 6.6) at the plasmalemma depends on the distribution of cortical F-actin in chromaffin cells, (Torregrosa-Hetland et al., 2011).

Furthermore, the cortical cytoskeleton can directly produce mechanical closure of membrane disruptions. In Xenopus oocytes, formation of a contractile actomyosin ring (AMR) constricts disruptions by generating tensile forces in the surrounding plasmalemma (Mandato and Bement, 2001). In the AMR, myosin-2 accumulates in a zone bordering the wound edge and F-actin localizes in the surrounding area (Benink and Bement, 2005). The Rho family of GTPases (Cdc42, Rac1, RhoA) act as regulatory proteins in the AMR. Active RhoA colocalizes to myosin-rich regions and phosphorylates the myosin light chain (MLC), increasing actomyosin contractibility. Similarly, Cdc42 associates with F-actin and dephosphorylates the MLC, decreasing contractability. These opposing interactions function to constrict disruptions and decrease the amount of membranous structures required to repair holes in Xenopus oocytes (see

Darenfed and Mandato, 2011 for review). The authors describe both biochemical processes and sub-cellular events, a rare correlation of biochemical pathways with micro-morphological events in most studies of plasmalemmal repair discussed in Section 6.

Cytoskeletal dynamics in plasma membrane repair are likely a conserved evolutionary process because cytoskeletal components are required for repair in many model systems. For example, F-actin is involved in sealing membrane disruptions in *Drosophila* embryos (Abreu-Blanco et al., 2012) and frog oocytes (Sokac et al., 2003; Ebstrup et al., 2021). However, there are inconsistencies in the precise role of cytoskeletal components. Taxol, an MT stabilizing agent, and cytochalasin E, an actin destabilizing compound, had no significant effect on sealing crayfish MGAs (Godell et al., 1997). In contrast, both substances inhibited sealing of mammalian septal axons (Xiea and Barrett, 1991).

Many aspects of the involvement of the cytoskeleton in membrane repair is not yet well-examined, such as the possible role of cytoskeletal proteins to transport membranous structures to sites of plasmalemmal/axolemmal damage. For example, MTs are disassembled and reassembled towards the membrane disruption (See Section 6.2; Togo, 2006) and may serve to better transport vesicles to the site of injury. Alternatively, vesicles may be carried towards the site of injury by the bulk flow of cytoplasm out of the plasmalemmal disruption.

6 Biochemical pathways of plasmalemmal sealing and their biophysical/morphological correlates

In this section, we discuss both well-established and recently identified biochemical pathways and some biophysical/ morphological mechanisms that contribute to the formation of a diffusion barrier (seal) at a lesion site.

Numerous parallel biochemical pathways contribute to the formation of a plasmalemmal seal after plasmalemmal lesions. These pathways are initiated by Ca2+ influx, along with (perhaps to a lesser extent) release of Ca2+ from internal sources (SER, mitochondria), reactive oxygen species (ROS) and mechanical distortion/stress of the plasmalemma (Cai et al., 2009a; Hendricks and Shi, 2014; Cheng et al., 2015). The substances, pathways, and subcellular organelles activated by traumatic injury all converge on one or more cellular biophysical/ morphological processes: wound stabilization, cvtoskeletal reorganization, and promotion of vesicle-vesicle and/or vesiclemembrane fusion (Figure 3). To date, the intermediate substances in these biochemical pathways are rarely correlated with observations of molecular/cellular events in the sealing process.

6.1 Calcium

Intracellular Ca²⁺ concentration ranges from 10^{-6} to 10^{-7} M in most eukaryotic cells (Bagur and Hajnóczky, 2017; Purves et al., 2018). Upon plasmalemmal damage, intracellular Ca²⁺ concentration rises primarily due to influx of extracellular Ca²⁺ at the lesion site along with possible release from intracellular stores. [see Chang and Reynolds (2006) for review]. This injury-dependent rise in intracellular Ca^{2+} initiates most biochemical pathways involved in plasmalemmal sealing, including calpain and other proteases, phospholipase, annexins, and adenylate cyclase (Figure 3). Once activated, these pathways ultimately promote vesicle-membrane interactions and cytoskeletal dynamics (Eddleman et al., 1997; Blanchette et al., 1999; Eddleman et al., 2000; Benink and Bement, 2005). Intracellular Ca²⁺ concentration must be raised (likely above 10^{-4} M) to induce sealing in cultured rat hippocampal neurites (Xiea and Barrett, 1991). The plasmalemmal repair process is initiated by this influx of Ca²⁺, rather than plasmalemmal damage *per se* (Bittner et al., 2016b). Therefore, the time course of sealing is properly measured post- Ca²⁺ influx (i.e., addition to media), rather than post-injury (Yoo et al., 2003).

Several substances have been reported to enhance sealing (or sealing processes) in the absence of Ca^{2+} . For example, overexpression of TRIM (MG53) in C2C12 cell culture results in the development of filopodia-like structures (Cai et al., 2009a), possibly representing the increased membrane surface area often seen in response to injury. Additionally, PEG (See Section 9) can induce artificial complete and rapid sealing in Ca²⁺-free solutions (Spaeth et al., 2012a; Spaeth et al., 2012b).

6.2 Phospholipase pathway

Ca²⁺ influx following injury activates several phospholipases that influence sealing through parallel biochemical pathways (Figure 3). Phospholipase C (PLC) and phospholipase A2 (PLA2) contain Ca²⁺sensing domains (Leslie and Channon, 1990; Clark et al., 1991), suggesting a role as initiating factors of the phospholipase response. Yawo and Kuno (1983) reported that inhibitors of PLA2 prevented sealing of transected cockroach giant axons. Activated PLA2 yields arachidonic acid (AA) and lysophosphatidic acid (LPA) through cleavage of lysophospholipid and phospholipid, respectively (Hendricks and Shi, 2014).

Novel PKC (nPKC) and conventional PKC (cPKC) are stimulated by AA (Liu and Heckman, 1998; Obis et al., 2015). Injured astrocytes in culture release AA extracellularly for several hours after wounding (Lamb et al., 1997). PKC isoforms (nPKCŋ and nPKCθ) activate phospholipase D (PLD) (Han et al., 2002; Spaeth et al., 2010). Additionally, both nPKC and cPKC activate myristoylated alaninerich C-kinase substrate (MARCKS) (Laux et al., 2000). MARCKS can release PIP2 from the plasmalemma, which is cleaved by PLC to yield DAG and IP3 (Goni et al., 2012). IP3 increases cytoplasmic Ca²⁺ by binding to receptors that induce release of intracellular Ca2+ stores. PLD is activated by ADP-ribosylation factor (Arf)-1 (Hodgkin et al., 1999), a Rho-family GTPase. PLD releases phosphatidic acid (PA) from phosphatidylcholine (PC) (Hendricks and Shi, 2014). PA then forms DAG through by stimulation of phosphatidylinositol-4-phosphate 5kinase (PI(4)P 5-Kinase) (increasing PIP2 concentrations) or by direct cleavage through PA hydrolase (Roth, Bi, Ktistakis, and Yu, 1999). LPA released by PLA2 activates Rho signaling. PLC cleaves PIP2, a link between cortical actin and the plasmalemma (Raucher et al., 2000). PIP2 cleavage by PLC may decrease membrane tension by disruption of contractile actin filament arrays, an effect on subcellular structures that has not yet been studied to confirm a correlation of such biochemical and subcellular structural events.



FIGURE 3

Biochemical pathways involved in natural and artificial repair of plasmalemmal disruptions. Membrane disruption allows an influx of Ca^{2+} into the cell due to the transmembrane concentration gradient of Ca^{2+} . Increased intracellular Ca^{2+} activates several parallel pathways involving annexins, oxidative responses, phospholipases, calpains (and other proteases), cAMP, and dynamic changes to membrane lipid compositions. ESCRT proteins may participate in the closure of small nm-sized holes that may involve membrane externalizations. Solid black lines and boxes outlined in black describe general pathways. Boxes outlined in blue describe the involvement of multiple cytoplasmic substances. Boxes outlined in red describe sub-cellular biophysical/morphological events. All these naturally-occurring pathways, cytoplasmic substances, and subcellular organelles converge onto three cellular biophysical/morphological events common to membrane sealing/repair in all eukaryotic cells: stabilization of the membrane surrounding the disruption, reorganization of cytoskeletal elements, and promotion of vesicle-vesicle and/or vesicle-membrane interactions. Finally, polyethyleng glycol (PEG) is an artificial fusogen that bypasses all known Ca^{2+} -dependent natural sealing pathways to rapidly repair plasmalemmal disruptions (PEG-sealing).

6.3 Oxidation pathway

Upon plasmalemmal disruption, reactive oxygen species (ROS) are produced by, or are affected by, cytoplasmic reactions and organelles (Spaeth et al., 2012). For example, the mitochondrial Ca²⁺ uniporter takes up excess Ca²⁺ following wounding. Increased Ca²⁺ produces a transient increase in the mitochondrial production of ROS (Cai et al., 2009a; Cai et al., 2009b). An increase in ROS activates the GTPase RhoA, triggering the accumulation of F-actin at the injury site (Figure 3). Inhibition of the mitochondrial Ca²⁺ uniporter or ROS significantly decreases plasmalemmal sealing in mouse C2C12 myoblasts (Horn et al., 2017). RhoA recruits and activates both mDia1 and Rho-associated protein kinase (ROCK). The activated mDia1 promotes actin polymerization while ROCK phosphorylates the MLC, increasing actomyosin contractibility (Boyer et al., 2006; Sarasa-Renedo et al., 2006). The interactions of mDia1 and ROCK are identical to those in the formation of a contractile AMR (Section 6.2), suggesting redundancy in the regulatory signaling involved in plasmalemmal repair.

ROS also regulate pathways that yield vesicle-membrane interactions similar to those activated by Ca2+. In cultured muscle cells, activation of Tri-Partite Motif (TRIM) family proteins is oxidation-dependent (Cai et al., 2009a). TRIM72 (also known as MG53, a muscle-specific protein) accumulates at phosphatidylserine (PS)-rich regions of the plasmalemma and on the surfaces of intracellular vesicles, alongside Dysferlin and Caveolin-3 (Cav3) (Cai et al., 2009a). Activation of TRIM family proteins by ROS enhances sealing through interaction with synaptic fusion proteins (Spaeth et al., 2012c). Additionally, ROS induce surface-bound MG53 oligomerization (Boucher and Mandato, 2015) and thus may enhance the tethering of vesicles to each other and to the plasmalemma during plug formation (i.e., facilitating vesicle-vesicle interactions). Specifically, mini-dysferlin72 associates with MG53 at the plasmalemma disruption, suggesting involvement of calpainmediated cleavage of full length dysferlin (Lek et al., 2013). Crosslinking of vesicles and/or the membrane via MG53 oligomerization serves as a prime example of a

biochemical interaction being related directly to cytoplasmic biophysical/morphological mechanisms of plasmalemmal repair.

6.4 Calpains

Calpains (Figure 3) are a family of Ca²⁺-dependent proteases that cleave cytoskeletal substances that then release membrane bound PLC and PKC. Calpain-activated PLC increases diacylglycerol (DAG) and IP3 through cleavage of PIP2 (Zuzek et al., 2012). PKC increases intracellular [DAG] by phosphorylating MARCKS. Activation of endogenous calpains by elevated cytoplasmic Ca²⁺ concentration enhances proteolysis of cortical cytoskeleton proteins (Siman, Noszek, and Kegerise, 1989). Targets of Ca2+-activated calpain include dysferlin, AHNAK, and spectrin (Huang et al., 2008). Axons express a network of F-actin filaments interconnected by an association with spectrin parallel to the plasmalemma (Xu, Zhong, and Zhuang, 2013) which could act as a barrier to endocytic events. Hence, cleavage of cytoskeletal linking proteins by calpains may enhance plasmalemmal sealing by decreasing membrane tension, similar to the action of PLC (Section 6.2).

Similar to their role in regulating actin dynamics, calpains target MT-associated proteins (Johnson, Litersky, and Jope, 1991). Talin, which interacts with actin capping proteins to promote cytoskeletal reorganization, is cleaved by calpain (Camussi et al., 2010). Calpain may be necessary for sealing since transected squid giant axons do not completely seal when monitored up to 60–120 min post-transection without the addition of exogenous calpain (Godell et al., 1997). Similarly, the sealing of transected crayfish MGAs is impaired by inhibitors of exogenous calpain. Deletion of calpain-encoding genes in mouse embryos decreases their ability to seal (Mellgren et al., 2007).

6.5 Annexins

Annexins are a family of proteins (Figure 3) that exhibit Ca²⁺dependent binding to negatively charged membrane phospholipids (PS, PA, and PI) (reviewed by Gerke et al., 2005; Boye and Nylandsted, 2016). Annexins show a graded response to cellular wounding and Ca²⁺ influx, with AnxA6 and AnxA2 being the most sensitive to Ca²⁺ and earliest acting in the sealing process. AnxA1 and AnxA5 are activated later when intracellular Ca2+ concentration is increased (Rosengarth et al., 2001; Monastyrskaya et al., 2007). Following Ca²⁺-influx, AnxA1, and AnxA2 hetero-tetramerize with S100 proteins. The AnxA2/S100A10 complex associates into a multimeric complex with AHNAK in a Ca²⁺-independent manner (Benaud et al., 2004) and binds dysferlin localized at the injury site (Huang et al., 2007). AnxA2 associates with the actin bundles that tether secretory granules to the plasmalemma in chromaffin cells (Gabel and Chasserot-Golaz, 2016). This complex may enhance plasma membrane stabilization and/or actin cytoskeletal remodeling (Demonbreun et al., 2016).

Similarly, AnxA5 at membrane edges surrounding the wound site associates with PS to form a stable, two-dimensional array that may prevent wound expansion in oocytes (Bouter et al., 2011). AnxA5 is required for sealing of human myotubes (Carmeille et al., 2016). The role of AnxA5 in wound stabilization is an important example of a correlated biochemical and cellular events lacking for most substances in various pathways.

Membrane blebbing has been proposed as an alternate mechanism to seal small (<100 nm) plasmalemmal holes (see Koerdt, Ashraf, and Gerke, 2019 for review). Membrane blebs incorporate injured membrane into a vesicle-like structure that is sealed off from the rest of the cytoplasm by an AnxA1 plug (Draeger, Monastyrskaya, and Babiychuk, 2011). The cellular utilization of this mechanism other than for non-traumatic, toxin-induced, pore expression is yet unknown.

6.6 cAMP pathway

Increased intracellular Ca²⁺ concentration upregulates cyclic AMP (cAMP) that activates both PKA and Epac1/Epac2 (Spaeth et al., 2012a; Spaeth et al., 2012b; Spaeth et al., 2012c; Zuzek et al., 2012; Togo, 2017). Activated PKA increases the activity of downstream synaptic fusion proteins such as synaptotagmin VII and SNAP-25 (Yoshihara, Suzuki, and Kidokoro, 2000). In a parallel pathway (Figure 3), the guanine exchange factors (Epac1 and 2) facilitate activation of both synaptic fusion proteins and Golgi fusion proteins. Epac produces an increase in neurotransmitter release in rat hippocampal (Ster et al., 2009), PC12 (Hatakeyama et al., 2006), and B104 cells (Spaeth et al., 2012c). Epac also activates Rap1, an upstream effector of Cdc42 (de Rooij et al., 1998). Both Rap1 and Cdc42 accumulate at PIP2 and PIP3 rich portions of the plasmalemma (Johnson, Erickson, and Cerione, 2012). This accumulation ultimately serves to localize dynamic branched F-actin adjacent to the disruption through activation of neutral Wiskott-Aldrich syndrome protein (N-WASP) and then Arp2/3 proteins (Rohatgi et al., 1999). This evolutionarily-conserved mechanism of branched actin formation regulates membrane tension and curvature, as well as facilitates clathrin-mediated endocytosis (for review see Papalazarou and Machesky, 2021). Branched action filaments are mechanosensitive, perhaps aiding in the initiation of repair processes in response to stretch injuries.

6.7 DAG

Activation of the phospholipase and calpain pathways increases the localized concentration of DAG in raft sections of a plasmalemma. DAG activates proteins involved in neurotransmitter release (Zuzek et al., 2012), perhaps by reducing hydration of bilayer surface substances (Goni and Alonso, 1999) and facilitating vesicle fusion and/or vesicle budding (Hendricks and Shi, 2014). DAG activates cPKC and nPKC by interacting with their C1 domain (Liu and Heckman, 1998; Goni et al., 2012). DAG can be metabolically altered to produce several membrane phospholipids, including PS and PC, that affect membrane stabilization and curvature (Millan-Oropeza, 2017). Finally, DAG produces vesicle exocytosis *via* its interactions with Munc13-1 and Doc2a, two synaptic fusion proteins (Mochida et al., 1998). Much of DAG's role in membrane repair involves altering membrane-specific properties, such as membrane curvature. However, *in vivo* evidence is lacking for any advantage conferred by DAG in the membrane repair processes, such as enhanced vesicle formation or specific downstream interactions.

6.8 ESCRT

Endosomal Sorting Complex Required for Transport (ESCRT) accumulates at sites of nanometer-sized plasmalemmal holes until sealing is complete (Jimenez et al., 2014) and is one of the few examples where a substance in the sealing pathway has its biochemical pathway correlated with specific biophysical/ morphological cellular effects (Table 2; Figure 2). The ESCRT machinery is assembled at the site of injury, initiated by the Apoptosis-Linked Gene 2 (ALG-2)-a Ca²⁺-binding protein. ALG-2 associates with specific ESCRT subunits beginning with ALG-2-Interacting protein, followed by ESCRT-III and Vps4 (Scheffer et al., 2014). AnxA7 may help recruit ALG-2 to a site of plasmalemmal damage (Cai et al., 2009a). Filaments composed primarily of ESCRT-III form at the plasmalemma and promote topographic changes in the membrane followed by scission (reviewed by Schoneberg et al., 2016). ESCRT complexes seal small (<100 nm wide) plasmalemmal holes generated by mechanical or laser-induced micro-punctures (Jimenez et al., 2014), or by pore-forming toxins, which produce stable protein multimers that span the plasma membrane. Given the small diameters of these plasmalemmal wounds and the role of ESCRT proteins and adaptors in endosomal processes, ESCRT proteins may have auxiliary roles to help seal complete transections or larger plasmalemmal holes caused by traumatic injury (Moe et al., 2015).

6.9 NSF

Pathways facilitating activation of both Golgi and synaptic fusion proteins ultimately converge onto N-ethylmaleimide (NEM)-sensitive factor. Activation of oxidation, PKA, and Epac pathways simultaneously cannot overcome inhibition of NSF by NEM treatment (Spaeth et al., 2012a; Spaeth et al., 2012b; Spaeth et al., 2012c; Zuzek et al., 2012), confirming that NSF acts downstream of these sealing pathways in B-104 cells. NEM also inhibits NSF-induced fusion of membranes in rat neurites (Garcia et al., 1995).

6.10 Lysosomal and enlargeosomal exocytosis

Lysosomal exocytosis may function as an initial response to wounding by expanding the pool of available vesicles for plasmalemmal sealing. Lysosomes fuse with the plasma membrane following an influx of Ca²⁺. Acid sphingomyelinase generates ceramide by cleavage of sphingomyelin, which promotes negative membrane curvature (Holopainen, Angelova, and Kinnunen, 2000) and thus inward budding of the plasmalemma (Tam et al., 2010; Corrotte et al., 2013). Cav1 or Cav3 are required for this endocytic internalization (Corrotte et al., 2013). Both Cav3 and dysferlin are downstream targets of TRIM family proteins/MG53 (Section 6.3). In adult muscle cells, dysferlin is rapidly recruited to the site of injury in an actincytoskeleton-dependent manner and is required for endocytosis (McDade, Archambeau, and Michele, 2014). Exocytosis of lysosomes involves synaptotagmin VII (Martinez et al., 2000), a synaptic fusion protein which binds syntaxin at intracellular Ca^{2+} levels considerably lower than other synaptogamin isoforms (Krasnov and Enikolopov, 2000). Given the low Ca^{2+} concentration dependency of many fusion proteins, this response is likely initiated rapidly and functions early in the sealing timecourse.

Another rapid initial response to wounding may involve the (<100 nm exocytosis of small diameter) vesicles (enlargeosomes). First identified in PC12 clones (Borgonovo et al., 2002), enlargeosomes fuse to the plasmalemma, thereby increasing the surface area and in turn decreasing the membrane surface tension. Enlargeosomal fusion may be mediated by synaptic SNARE substances including VAMP4 (Cocucci et al., 2008). AnxA2 has also been implicated in this specific exocytotic process (Chasserot-Golaz et al., 2005). Enlargeosomes would function to increase the amount of membrane area available for repair vesicles and to decrease membrane tension.

6.11 Removal of pore-forming toxins

In addition to mechanically-induced membrane wounds, the plasmalemma can be disrupted by the insertion of stabilized membrane pores. Bacterial toxins such as streptolysin O damage cells through the insertion of large transmembrane pores that induce cell death and/or nutrient release (reviewed in Boucher et al., 2019). These pore-forming toxins (PFTs) allow for uncontrolled influx of Ca^{2+} into the target cell.

Eukaryotic cells do not repair these injury types by the accumulation of vesicles, but rather by the excision of surrounding membrane (Iacovache et al., 2012). PFTs are removed by either shedding of membrane blebs or endocytic degradation. Ca^{2+} influx through PFTs activates biochemical pathways that are shared with vesicle-mediated membrane repair. Formation of membrane blebs is promoted by calpain-mediated disruption of the cortical cytoskeleton (Atanassoff et al., 2014) and/ or selective insertion of PFTs into cholesterol-rich regions of the plasmalemma resulting in changes to membrane curvature (Drucker et al., 2018). PFT removal involves the recruitment of annexins to PFT sites that produce cytoskeletal changes that contract the neck of forming membrane blebs (Babiychuk et al., 2009; Atanassoff et al., 2014).

Lysosomal endocytosis (see Section 6.10) may also be involved in the removal of PFTs from the plasmalemma. PFT permeabilization results in increased sphingomyelinase activity, favoring negative membrane curvature through the generation of ceramide in the plasmalemma (Walev et al., 2001). The endocytotic mechanism of PFT removal of inward membrane curvatures is reported to occur *via* active Ca²⁺-dependent processes and/or *via* passive Ca²⁺-independent processes (Boucher et al., 2019). PKC and p38 MAPK is upregulated upon membrane disruption by PFTs (Stassen et al., 2003), perhaps further directing changes in local membrane composition through the downstream effect on the DAG pathway (see Section 6.7).

7 Current models of plasmalemmal sealing

The pathways and cellular events described above and in Figures 2, 3 are consistent with vesicle-mediated, biophysical/morphological sealing mechanisms in which the influx of Ca²⁺ through the lesion site causes the formation of membranous structures that form a diffusion barrier at the lesion site. Two biophysical/morphological mechanisms (patch and plug) have been proposed to explain the initial seal (see Figure 2). These two hypothetical mechanisms involve the same initial vesicular interactions and biochemical pathways, but differ in the final events that form a complete functional barrier that has the diffusion characteristics of the original intact plasmalemma (Bittner and Fishman, 2000; Davenport et al., 2016) but not its morphological structure (Lichstein et al., 2000). In fact, the two biophysical/morphological sub-cellular mechanisms may be distinctions without a difference in terms of outcome, given that both ultimately form a diffusion barrier that approximates the permeability and ultrastructure of an undamaged plasma membrane. In the following subsections, we describe each model and examine the evidence for and against both hypothesized biophysical/morphological mechanisms.

7.1 The vesicle patch model

As proposed for Xenopus oocytes and sea urchin eggs, the vesicle patch model hypothesizes a one-step membrane sealing mechanism in which vesicles accumulating at the wound site first fuse together to form a large membranous "patch" which then completely repairs the plasmalemmal hole or transection in a single fusion step. This theoretical biophysical/morphological mechanism is based primarily on an observation that Ca2+-containing sea water injected into a sea urchin egg is quickly contained within a newly formed "wound vesicle" (Terasaki et al., 1997). A good schematic figure for this patch mechanism (Davenport et al., 2016) is derived from indirect data that are somewhat controversial (Bittner and Fishman, 2000; Bittner et al., 2016b; Vargas and Bittner, 2019). For instance, McNeil and Baker (2001) show several images of limited vesicle-vesicle fusion events-but do not publish an example of a complete wound patch. Davenport et al. (2016) proposes the process of "explodosis" in which the outer membrane layer of the patch opens to yield a single bilayer, shown only in a schematic figure. This cellular event has not been directly observed and published.

If a plasmalemmal disruption seals in a single step, one would expect to see a stepwise change in V_{m} , R_{in} , I_i , and dye exclusion. However, all reports of these functional measures indicate a progressive and gradual change in other sealing models that do not have an abundance of pre-formed, sub-plasmalemmal vesicles (Bittner and Fishman, 2000; Vargas and Bittner, 2019). Furthermore, the rate and extent of dye exclusion depends inversely on the dye molecular size in transected crayfish MGAs, consistent with a progressive formation of a seal (Eddleman et al., 2000).

7.2 The vesicle plug model

In contrast to a single-step patch seal by a wound vesicle, the vesicle plug hypothesis assumes that plasmalemmal holes or complete transections of cytoplasmic extensions are sealed by an accumulation of vesicles which more gradually forms a diffusion barrier through vesiclevesicle and vesicle-plasmalemmal interactions. Vesicles and other membrane bound structures (single and multilayered vesicles, lysosomes, mitochondria, etc.) interact with each other and the plasmalemma at the lesion site to form a "vesicular plug," that form a diffusion barrier to particles of smaller and smaller molecular size. Several biophysical/morphological mechanisms likely play a role in the formation of a vesicle plug, including:

- a) Dense packing of vesicles at the site of injury creates a longer and narrower diffusion path, possibly involving vesicle to vesicle attachment proteins such as gap junctions, adherens, and desmosomes (Bittner and Fishman, 2000; Fishman and Bittner, 2003):
- b) Contraction of an actomyosin ring (See Section 6.2) surrounding the plasmalemmal disruption may account for the observed narrowing of transected GAs (Krause et al., 1994; Bittner and Fishman, 2000; Eddleman et al., 2002). Additionally, contraction is suggested to help seal small plasma membrane holes (Mandato and Bement, 2001; Benink and Bement, 2005). However, membrane contracture has not been demonstrated to yield complete membrane closure/sealing.
- c) Internalization or budding of membrane, mediated by ESCRT. Endocytosis could remove and subsequently seal small (<100 nm) plasmalemmal disruptions (Jimenez et al., 2014; Moe et al., 2015).
- d) Fusion of vesicles to form larger or even multi-layered vesicles that would continue to accumulate at the wound site. In crayfish MGAs, squid GAs, and earthworm GAs, plasmalemmal injury induces formation of single-walled vesicles of plasmalemmal origin and multi-walled vesicles likely of glial origin (Ballinger et al., 1997; Eddleman et al., 1998a).

Via a combination of these events and vesicle accumulation, a vesicular plug is hypothesized to progressively form a diffusion barrier at the plasmalemmal lesion (Krause et al., 1994; Eddleman et al., 1997; Eddleman et al., 1998a; Eddleman et al., 2000; Fishman and Bittner, 2003) The following results are consistent with the vesicle-plug hypothesis.

- a. Morphological observations show a gradual accumulation of membranous structures at sites of disruption associated with a diffusion barrier to smaller and smaller molecules (Fishman and Bittner, 2003) rather than a one-step sealing event. SEM observations of squid GAs and crayfish MGAs show a nonuniform accumulation of vesicles at the cut end of axons (Eddleman et al., 2002). Furthermore, SEM of unfertilized sea urchin eggs reveals heterogeneous massing of membranous structures following injury (McNeil and Baker, 2001).
- b. Sealing is slower in MGAs and other cells that do not have large populations of pre-formed vesicles directly below the plasmalemma such as *Xenopus* oocytes and unfertilized sea urchin eggs (Bittner G. D. et al., 2000; Fishman and Bittner, 2003). These numerous vesicles in unfertilized eggs rapidly migrate to the injury site and form a diffusion barrier within seconds (Steinhardt et al., 1994; Terasaki et al., 1997).
- c. The sealing of smaller lesions is more rapid than larger lesions. PC12 cells (Yoo et al., 2004) and the neurites of B104 cells

10.3389/fphys.2023.1114779

(McGill et al., 2016) show much slower sealing times than that of crayfish MGAs and squid GAs (Bittner and Fishman, 2000; Fishman and Bittner, 2003).

- d. The time to exclude tracer molecules varies inversely with molecular weight and molecular diameter of the tracer. Larger fluorescent dye molecules placed extracellularly are initially excluded followed by the exclusion of tracers of smaller and smaller size (Eddleman et al., 2000). Furthermore, ions of yet-smaller molecular weight and size are excluded later than small fluorescent dyes of higher molecular weight (Eddleman et al., 1998a).
- e. Probability of sealing varies monotonically and continuously with the time at which Ca^{2+} influx is initiated. When cells are transected in Ca^{2+} -free physiological saline and then placed in a Ca^{2+} -containing solution, dye barriers form at similar post-calcium addition times rather than at similar post-transection times (Yoo et al., 2004). Furthermore, sealing time increases with distance of the lesion site from the soma (Yoo et al., 2004; Nguyen et al., 2005).

Together, these results provide evidence against a one-step sealing event that should be produced by a wound vesicle. In contrast, these results are consistent with the vesicle plug hypothesis, in which vesicle accumulation followed by vesiclevesicle and/or vesicle-membrane interactions progressively form a seal.

Future research to resolve this controversy could use more powerful current microscopy and labeling techniques. Recentlyenhanced time-lapse confocal microscopy and threedimensional reconstruction of images should reveal mechanisms of vesicle accumulation and interaction in various cell types (including invertebrate GAs and unfertilized oocytes). Modified SEM techniques that eliminate artifacts due to oxidation during fixation (Chuo et al., 2018) could distinguish between patch vs plug mechanisms of vesicle accumulation. The combined use of current confocal and SEM imaging could also determine if the mechanism of seal formation is universal among all eukaryotic cells or is cellspecific, i.e., determined by cell morphology and the immediate availability of membranous structures to seal a damaged plasmalemma.

7.3 Reestablishment of a continuous lipid bilayer

Formation of a vesicle seal—either through the patch or plug hypothesis—results in a membrane topology very different from that of its uninjured state. SEM images of both crayfish MGAs (Eddleman et al., 2002) and unfertilized *Xenopus* oocytes (McNeil and Baker, 2001) show collections of variously-sized membranous structures ($0.5-6 \mu m$ on average) jutting out from the injury site. Furthermore, successful vesicle-mediated repair changes the spatial distribution of lipid species and proteins in the plasmalemma. Following injury, microdomains (rafts) enriched with specific lipid species such as DAG and sphingomyelin form to facilitate repair processes (see Section 6.7 and Section 6.10; Holopainen, Angelova, and Kinnunen,

2000; Hendricks and Shi, 2014). It is likely that the distribution of membrane-associated proteins is similarly disrupted following injury. Thus, cell wounding and subsequent vesicle-mediated repair results in a diffusion barrier with a vastly different topology and distribution of proteins and other substances compared to its native (uninjured) state.

The native distribution of lipid species throughout the plasmalemma must presumably be eventually re-established following initial seal formation (Sønder et al., 2021). While a continuous membrane bilayer resembling that of the uninjured cell is observed at 24 h following injury (Lichstein et al., 2000), no long-term imaging studies of injured cells have been performed to this date. Reestablishment of the native membrane lipid composition is also currently unknown and might occur by (1) lateral flow of vesicle-lipids into the seal surround as vesicles begin to collapse into the plasmalemma or (2) the *de novo* synthesis and integration of membrane lipids into the plasmalemma in conjunction with removal and breakdown of repair vesicles. New techniques in *de novo* lipid labeling combined with time-lapse confocal microscopy could be used to determine the contribution of lipid metabolism to restoration of a membrane bilayer.

8 Adaptive responses following a conditioning plasmalemmal sealing event

Uncontrolled Ca^{2+} influx at sites of membrane disruption produces rapid degeneration of cellular structures and eventually cell death (Schlaepfer and Bunge, 1973; Trump and Berezesky, 1995). The ability of a cell to retard this Ca^{2+} influx and re-form a membranous barrier to the extracellular environment is crucial to its survival. As described in previous sections, eukaryotic cells repair plasmalemmal lesions and prevent apoptosis by evolutionarily conserved subcellular mechanisms using vesicular interactions and biochemical pathways. This section describes adaptive responses following plasmalemmal injuries that are common to many eukaryotic cell types that enable them to prevent further injury and/or more rapidly seal a second plasmalemmal injury.

After wounding, cells display short-term and long-term adaptations to respond more rapidly to future wounding in the damaged cell and in neighboring undamaged cells (reviewed in Togo, 2019). Short-term responses can involve an increase in endocytosis, expansion of vesicle pools, and migration of available membranous structures to plasmalemmal lesion sites. Long-term responses involve the activation of transcription factors that regulate cell survival. Figure 4 summarizes the primary biochemical pathways involved in the diverse set of adaptive responses activated following the successful formation of a membrane seal to increase cell survival in the event of a subsequent membrane injury.

MDCK cells wounded for a second time in tissue culture at the same site of a previous injury exhibit resealing rates nearly double that of cells wounded a single time (Togo, 2012), i.e., a facilitated short-term adaptive response. This short-term adaptive response activates PKC, possibly leading to phosphorylation of proteins involved in exocytosis, similar to those depicted in Figure 3



cells exhibit short- (black lines) and long-term (red lines) potentiation of membrane repair processes. In both single cells and populations of injured cells, long-term potentiation involves transcription factor-mediated upregulation of genes associated with plasmalemmal repair and cell survival. Note that many biochemical pathways and substances are common to plasmalemmal repair of traumatic lesions to single cells (see Section 6) and adaptive response pathways in the same cell or neighboring cells.

(Togo et al., 1999; Togo et al., 2003). This short-term facilitation is sensitive to Brefeldin A, suggesting the involvement of Golgi fusion proteins. The facilitated response also exhibits the reorganization of microtubules (Togo, 2006; see Section 6.2 for further details).

This potentiated short-term resealing response occurs when a cell is wounded at a second site at least 10–20 μ m from the initial injury site. This short-term potentiated response requires PKA, but not PKC (Togo et al., 1999; Togo et al., 2003; Togo and Steinhardt, 2004). ATP is released extracellularly either by exocytosis or diffusion through the membrane disruption (Togo, 2017). Extracellular ATP activates P2Y G-proteincoupled receptors and P2X ligand-gated ion channels (Burnstock, 2007). P2Y receptors cause a release of Ca²⁺ from intracellular stores (via PLC and IP3), increased PKC activity, and increased cytosolic cAMP concentrations (Balboa et al., 1994; Post et al., 1998). In contrast, activated P2X receptors produce a direct influx of Ca2+. The primary effect of these receptors is to activate the cAMP pathway. cAMP activates PKA and a subsequent increase in injury-induced vesiculation (Togo, 2019).

These short-term responses in wounded MDCK cells are also seen in neighboring cells within 5 min of the initial injury event (Figure 4). When injured, neighboring cells seal faster than the initially injured cell (Togo, 2014). This potentiation is observed up to 6 cells distant to the initially wounded cell (Togo, 2012) and is primarily attributed to the extracellular release of ATP from the initially wounded cell. Inhibition of extracellular ATP signaling eliminates the potentiated response in neighboring cells (Togo, 2017). ATP activates both a DAG and cAMP pathway in neighboring cells *via* binding to P2Y receptors, similar to pathways observed in non-adaptive repair (Section 6.6 and Section 6.7). These pathways activate PKC and PKA, yielding a short-term potentiated repair response in locally neighboring cells. Diffusion of calcium through gap junctions between wounded and neighboring cells may also induce potentiation.

Loading cells with styryl dye reveals an important difference between facilitated and potentiated short-term responses. Upon initial membrane disruption, localized vesiculation is primarily observed immediately surrounding the disruption. Wounding twice at the same site produces site-specific (local) vesiculation while subsequent wounding at a different site yields global vesiculation throughout the cell (Togo et al., 2003). The initial wounding of a cell activates several signaling pathways and increases the cell's capacity to seal additional membrane disruptions by the rapid formation of membranous structures.

In addition to the short-term adaptive resealing responses in wounded cells, long-term adaptations occur in both wounded and neighboring cultured MDCK cells. Both long- and short-term responses involve the phosphorylation of CRE-binding protein (CREB) that alters transcription. In 3T3 fibroblasts, membrane disruption stimulates PKC cascades. PKC then activates a p38 MAPK pathway that ultimately leads to CRE-mediated gene expression (Togo, 2004). PKC and p38 MAPK are also upregulated upon membrane disruption by pore-forming toxins (Stassen et al., 2003). Long-term responses may involve Akt/ PKB which has numerous downstream effectors regulating cell survival (see Song, Ouyang, and Bao, 2005 for review). Akt/PKB activates CREB to stimulate transcriptional control of apoptosis-related genes. Akt/PKB is localized to the cytoskeleton upon membrane disruption to interact with actin and Cdc42 (Cenni et al., 2003; see Section 6.2). Consistent with this result, injured cells often activate c-fos, a target of phosphorylated CREB (Sassone-Corsi et al., 1988; Grembowicz et al., 1999).

Long-term adaptive responses are also observed in neighboring cells. Diffusible signaling molecules such as nitric oxide (NO) are released from wounded cells (Togo, 2012). NO/ PKG signaling stimulates potentiation of membrane repair in neighboring cells. PKG stimulates CREB phosphorylation through ERK (Ciani et al., 2002). Similarly, extracellular ATP can stimulate ERK through activation of P2Y receptors (Neary et al., 2005). ATP-dependent activation of Y2X receptors can phosphorylate CREB through a PKC and CaMK pathway (Tan et al., 1996). ESCRT machinery has been proposed to generate extracellular vesicles that play a role in intercellular signaling following membrane disruption (Scheffer et al., 2014). Once activated, CRE in neighboring cells functions in a very similar manner to that described in a singly-wounded cell, i.e., primarily to regulate long-term potentiation of membrane repair. Other adaptive capabilities may include regulation of cell migration, survival, and regeneration (see Togo, 2019).

Although relatively few adaptive membrane repair responses have been studied, the adaptive pathways and substances involved are similar to those reported for the repair of single plasmalemmal lesions. The biochemical pathways reported have not yet been correlated with observations of sub-cellular biophysical/ morphological events, as is also the case for many single cell traumatic studies reported in previous sections.

9 Artificially-induced plasmalemmal repair

In addition to the natural methods of plasmalemmal sealing described in Sections 5–8, cell membranes can be fused or sealed by PEG, a 1–5 kDa synthetic amphoteric polymer that is commonly used to fuse cell membranes that are brought into close apposition (nanometers). PEG of low molecular weight (0.2–7 kDa) causes an osmotic imbalance across the plasmalemma that may alter membrane curvature and induce mechanical stress (Hermann et al., 1985; Money, 1989; reviewed in Malinin et al., 2002). However, this effect of PEG has only been investigated in the context of artificial vesicle systems (Mahendra et al., 2019), not *in vitro* or *in vivo*. Solutions of 1–5 kDa PEG have been used for decades to generate cell hybrids *in vitro* (Anne

and Peberdy, 1976; Yang and Shen, 2006) and to repair transected axons *in vitro* (Bittner et al., 1986; Krause and Bittner, 1990; Lore et al., 1999) and *in vivo* (reviewed in Bittner et al., 2016a; Bittner et al., 2016b; Van Nest et al., 2021; Bittner et al., 2022). More recently, PEG has been used as part of a therapeutic strategy to repair transected axons in clinical settings (Bamba et al., 2016; Bittner et al., 2022; Lopez et al., 2023).

PEG-fusion repair of singly-cut or crushed ends of earthworm giant axons (Krause and Bittner, 1990), rat sciatic and spinal axons (Lore et al., 1999) rapidly (ms to seconds) restores conduction of action potentials across transection sites and morphological continuity. PEG-fusion rapidly repairs severed open ends (presumed to be vesicle-free *in vivo*) of mammalian PNS axons brought into contact by neurorrhaphy and immediately restores involuntary muscle contractions. PEG-fused nerves exhibit return of voluntary behaviors after 2–6 weeks by inducing rewiring of PNS and CNS synaptic connections (Bittner G. et al., 2000; Bittner et al., 2016a; Bittner et al., 2016b; Ghergherehchi et al., 2019).

In the absence of PEG-fusion, PEG also has a neuroprotective effect on severed CNS axons (Liu-Snyder et al., 2007), perhaps by sealing off cut ends (PEG-sealing). In this model, application of PEG to the proximal or distal stump of a transected axon causes the rapid collapse and fusion of the axolemmal leaflets, similar to the previously assumed mechanism of axonal repair depicted in Figure 1B. That is, as indicated in biochemical and dye diffusion studies, PEG-sealing bypasses all substances and sealing pathways described in Section 6 (Spaeth et al., 2012c). PEG is able to seal severed axons in both calcium-free and calciumcontaining solutions and thus almost-certainly does not involve the formation of intracellular vesicles. Substances that are known to enhance (e.g., taxol) or inhibit (e.g., NEM, methylene blue) vesicle-mediated sealing had no measurable effect on PEG-sealing in B104 cells. Furthermore, the probability of successful PEGsealing increases sigmoidally as the concentration of PEG utilized increases from 0-10 mM, indicating that PEG directly induces sealing (Spaeth et al., 2012b).

10 Conclusions and future directions

Eukaryotic cells commonly experience plasmalemmal damage. For decades prior to 1994, the cellular/molecular mechanisms that repair plasmalemmal injuries were not discussed or were misinterpreted. Such undocumented theories of membrane repair included lipid spreading, plasmalemmal collapse and fusion, and the formation of membrane partitions. Since 1994, experimental studies consistently show that small or large plasmalemmal lesions seal by accumulation and aggregation of vesicles and/or other membranous structures. A combination of electrophysiological measures, fluorescent dye with series confocal Z-sections, and high-resolution EM images of the injury site are necessary to accurately determine the morphological and functional state of a plasmalemmal seal. Vesicles and membrane-bound structures interact with the plasmalemma and each other to produce a complete seal by forming (1) a vesicular plug consisting of many vesicles that gradually seal the membrane disruption or (2) a patch consisting of a large wound vesicle that seals the membrane disruption in a

Mencel and Bittner

single step. Both plug and patch sealing mechanisms assume that a set of parallel biochemical pathways and substances are primarily activated by the rapid influx of Ca^{2+} through the lesion site -- and perhaps secondarily by release of Ca^{2+} from intracellular structures (SER, mitochondria, etc.). These parallel biochemical pathways initiate less well-defined cellular events that facilitate membrane fusion and repair.

10.1 Cellular correlates of natural membrane sealing

Although many publications cited above provide detail on the biochemical pathways of plasmalemmal repair, few studies correlate subcellular morphological/biophysical events directly associated with biochemical substances. For instance, very little is known regarding the transport of wound-induced vesicles in cells. A "tether" that anchors vesicles to the plasmalemma has been described in crayfish MGAs (Eddleman et al., 1998a) perhaps pointing to delivery along cytoskeletal structures. Furthermore, MTs are disassembled and subsequently reassembled towards the injury site, perhaps to direct intracellular vesicle transport (Togo et al., 2000). Highresolution imaging of cells treated with various inhibitors of cytoskeletal filament polymerization and active transport are needed to evaluate the transport and aggregation of vesicles. Additionally, the role of fusogenic proteins in the repair process is not fully understood. While synaptic and Golgi fusion proteins are certainly involved in the repair process, the organization and overall contribution of these fusion complexes to vesicle aggregation is largely unknown.

Importantly, recent work has been successful in relating biochemical events with their cellular correlates. One example is the role of AnxA5 in wound stabilization. Annexins have long been known to interact with negatively charges phospholipids (such as PS) in a Ca²⁺-dependent manner (Gerke and Moss, 2002). However, it was only when high-resolution fluorescence imaging was performed on mouse perivascular cell culture that the cellular effect of AnxA5 was discovered. AnxA5 associates with PS-rich domains that surround the site of injury, forming rigid 2D crystals of AnxA5 trimers. AnxA5 arrays act to prevent expansion of the membrane injury and facilitate repair (Bouter et al., 2011). This discovery adds significant detail to our understanding of the repair process and demonstrates the importance of correlating the biochemical results to direct observations of the cellular biophysical/morphological events.

10.2 Clinical applications of natural membrane sealing

A better understanding of both natural mechanisms of plasmalemmal repair are almost-certainly critical to the development of new clinical treatments for conditions that involve damage to nerve cell membranes such as traumatic nerve injury, neurodegenerative diseases, muscular dystrophies, stroke and other ischemic conditions. Many neurodegenerative diseases are characterized by damage to cell membranes (reviewed in Dias and Nylandsted, 2021; Coumans et al., 2001). Muscle cells routinely experience membrane trauma (e.g., stretch injury) throughout their normal physiology. Plasma membrane instability characterizes many muscular dystrophies, brought about by mutations that compromise membrane repair and/or produce cells with fragile membranes (Davies and Nowak, 2006; Michailowsky et al., 2019). Dysregulations in annexin-family proteins, calpains, MG53, and dysferlin are involved in the development of muscular dystrophies (for review, see Croissant et al., 2021).

Mechanisms of the natural repair response may also have applications in the treatment of various cancers (Demonbreun and McNally, 2016). For example, Myoferlin-depleted Lewis Lung Carcinoma cells show a decrease in both migration and proliferation (Leung et al., 2013). Upon laser wounding, cells also show significant deficiencies in membrane repair capabilities. Myoferlin regulates endocytotic recycling/trafficking and is upregulated upon membrane injury, similar to dysferlin (Doherty et al., 2008; Demonbreun et al., 2010). Selective downregulation of ferlin-family member proteins may provide new avenues for novel cancer treatment.

10.3 Cellular correlates of artificial membrane sealing

PEG rapidly seals plasmalemmal lesions in milliseconds to seconds both *in vitro* and *in vivo* (Spaeth et al., 2012b; Vargas and Bittner, 2019). PEG bypasses all known substances involved in plasmalemmal repair studied to date (See Figure 3) to directly seal the lesion in a process, not yet directly observed, that is assumed to closely resemble the previously assumed mechanism of plasmalemmal collapse and fusion (See Section 4; Figure 1B). PEG might cause transected axonal leaflets to come together, fuse and seal the cut end within milliseconds to seconds, as compared with the natural methods of sealing that occur over seconds to minutes.

PEG-sealing has only been assessed by prevention of dye uptake and should now be studied more in depth using a combination of the methods discussed in this review. Individually transected crayfish MGAs can serve as a model system for studying PEGsealing. MGAs have a large enough diameter to facilitate confocal microscopy and electrophysiological measurement. Fluorescent dye combinations allow one to observe 1) the membrane dynamics (i.e., collapse and fusion), 2) the presence or absence of intracellular vesicles, and 3) the assessment of a seal by dye exclusion. These properties will allow for study of the cellular dynamics upon PEG addition. Additionally, many of the inhibitors of substances within the natural repair biochemical pathway have been used previously in crayfish MGAs (see Section 6) allowing one to study the dependency of PEG-sealing on elements such as the cytoskeleton, wound state, etc.

10.4 Clinical applications of artificial membrane sealing

PEG-sealing of may provide new avenues for the treatment of locally damaged cells and aid in the improvement of behavioral recovery following PEG-fusion of peripheral nerve injuries. Ischemic conditions (e.g., stroke) that are characterized by oxygen glucose deprivation in the cell often result in plasmalemmal damage and the formation of membrane blebs (Nishimura and Lemasters, 2001; Garcia-Dorado et al., 2012). Repair of ischemic cells by the natural repair response can be insufficient to prevent cell death and/ or degeneration of anucleate cytoplasmic extensions. The rapid sealing of holes and/or complete transections *via* local application of PEG could prevent apoptosis and improve behavioral recovery.

Author contributions

The authors confirm contribution to the paper as follows: MM and GB contributed equally to all aspects of this manuscript. All authors reviewed the results and approved the final version of the manuscript.

Funding

Funding was provided by the Lone Star Paralysis Foundation for research and publication costs.

References

Abe, N., and Cavalli, V. (2008). Nerve injury signalling. Curr. Opin. Neurobiol. 18 (3), 276–283.

Abreu-Blanco, M. T., Verboon, J. M., Liu, R., Watts, J. J., and Parkhurst, S. M. (2012). Drosophila embryos close epithelial wounds using a combination of cellular protrusions and an actomyosin purse string. *J. Cell Sci.* 125 (24), 5984–5997.

Alberts, B., Johnson, A., Lewis, J., Raff, M., Roberts, K., and Walter, P. (2015). *Molecular biology of the cell. Molecular biology of the cell.* 6th edn. New York: Garland: Science.

Andrews, N. W., Almeida, P. E., and Corrotte, M. (2014). Damage control: Cellular mechanisms of plasma membrane repair. *Trends Cell Biol.* 24 (12), 734–742. doi:10. 1016/j.tcb.2014.07.008

Anne, J., and Peberdy, J. F. (1976). Induced fusion of fungal protoplasts following treatment with polyethylene glycol. *J. General Microbiol.* 92 (2), 413–417. doi:10.1099/00221287-92-2-413

Atanassoff, A. P., Wolfmeier, H., Schoenauer, R., Hostettler, A., Ring, A., Draeger, A., et al. (2014). Microvesicle shedding and lysosomal repair fulfill divergent cellular needs during the repair of streptolysin O-induced plasmalemmal damage. *PLoS One* 9 (2), e89743. doi:10.1371/journal.pone.0089743

Babiychuk, E. B., Monastyrskaya, K., Potez, S., and Draeger, A. (2009). Intracellular Ca(2+) operates a switch between repair and lysis of streptolysin O-perforated cells. *Cell death Differ*. 16 (8), 1126–1134. doi:10.1038/cdd.2009.30

Bagur, R., and Hajnóczky, G. (2017). Intracellular Ca2+ sensing: Its role in calcium homeostasis and signaling. *Mol. Cell* 66 (6), 780-788. doi:10.1016/j.molcel.2017.05.028

Balboa, M. A., Firestein, B. L., Godson, C., Bell, K. S., and Insel, P. A. (1994). Protein kinase Ca mediates phospholipase D activation by nucleotides and phorbol ester in Madin-Darby canine kidney cells. Stimulation of phospholipase D is independent of activation of polyphosphoinositide-specific phospholipase C and phospholipase A2. *J. Biol. Chem.* 269 (14), 10511–10516. doi:10.1016/s0021-9258(17)34089-9

Ballinger, M. L., Blanchette, A. R., Krause, T. L., Smyers, M., Fishman, H. M., and Bittner, G. (1997). Delaminating myelin membranes help seal the cut ends of severed earthworm giant axons. *J. Neurobiol.* 33 (7), 945–960. doi:10.1002/(sici)1097-4695(199712)33:7<945::aid-neu6>3.0.co;2-8

Bamba, R., Waitayawinyu, T., Nookala, R., Riley, D. C., Boyer, R. B., Sexton, K. W., et al. (2016). A novel therapy to promote axonal fusion in human digital nerves. *J. trauma acute care Surg.* 81 (5), 177–183. doi:10.1097/TA. 000000000001203

Benaud, C., Gentil, B. J., Assard, N., Court, M., Garin, J., Delphin, C., et al. (2004). AHNAK interaction with the annexin 2/S100A10 complex regulates cell membrane cytoarchitecture. *J. Cell Biol.* 164 (1), 133–144. doi:10.1083/jcb.200307098

Benink, H. A., and Bement, W. M. (2005). Concentric zones of active RhoA and Cdc42 around single cell wounds. J. Cell Biol. 168 (3), 429-439. doi:10.1083/jcb. 200411109

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Supplementary Material

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fphys.2023.1114779/ full#supplementary-material.

Betz, W. J., and Bewick, G. S. (1992). Optical analysis of synaptic vesicle recycling at the frog neuromuscular junction. *Science* 255 (5041), 200–203. doi:10.1126/science.1553547

Biro, E. N., and Venyaminov, S. Y. (1979). Depolymerization of actin in concentrated solutions of divalent metal chlorides. *Acta biochimica biophysica; Acad. Sci. Hung.* 14 (1-2), 31–42.

Bittner, G. D., Bushman, J. S., Ghergherehchi, C. L., Roballo, K. C. S., Shores, J. T., and Smith, T. A. (2022). Typical and atypical properties of peripheral nerve allografts enable novel strategies to repair segmental-loss injuries. *J. Neuroinflammation* 19 (11), 60–27. doi:10.1186/s12974-022-02395-0

Bittner, G. D., and Fishman, H. M. (2000). Axonal sealing following injury. New York City: Nerve Regeneration: Marcel Dekker.

Bittner, G. D., Mikesh, M., and Ghergherehchi, C. L. (2016a). Polyethylene glycolfusion retards Wallerian degeneration and rapidly restores behaviors lost after nerve severance. *Neural Regen. Res.* 11 (2), 217–219. doi:10.4103/1673-5374.177716

Bittner, G. D., Schallert, T., and Peduzzi, J. D. (2000). Degeneration, trophic interactions, and repair of severed axons: A reconsideration of some common assumptions. *Neuroscientist* 6 (2), 88–109. doi:10.1177/107385840000600207

Bittner, G. D., Sengelaub, D. R., Trevino, R. C., Peduzzi, J. D., Mikesh, M., Ghergherehchi, C. L., et al. (2016b). The curious ability of polyethylene glycol fusion technologies to restore lost behaviors after nerve severance. *J. Neurosci. Res.* 94 (3), 207–230. doi:10.1002/jnr.23685

Bittner, G. D., Spaeth, C. S., Poon, A. D., Burgess, Z. S., and McGill, C. H. (2016c). Repair of traumatic plasmalemmal damage to neurons and other eukaryotic cells. *Neural Regen. Res.* 11 (7), 1033–1042. doi:10.4103/1673-5374. 187019

Bittner, G., Yoo, S., Fukuda, M., Nguyen, M. P., Bittner, G. D., et al. (2000). Axolemmal repair requires proteins that mediate synaptic vesicle fusion. *J. Neurobiol.* 44, 382–391. doi:10.1002/1097-4695(20000915)44:4<382::aid-neu2>3.0. co;2-q

Blanchette, M., Kunisawa, T., and Sankoff, D. (1999). Gene order breakpoint evidence in animal mitochondrial phylogeny. *J. Mol. Evol.* 4949 (22), 193–203. doi:10.1007/ pl00006542

Borgonovo, B., Cocucci, E., Racchetti, G., Podini, P., Bachi, A., and Meldolesi, J. (2002). Regulated exocytosis: A novel, widely expressed system. *Nat. Cell Biol.* 4 (12), 955–962. doi:10.1038/ncb888

Boucher, E., Goldin-Blais, L., Basiren, Q., and Mandato, C. A. (2019). Actin dynamics and myosin contractility during plasma membrane repair and restoration: Does one ring really heal them all? *Curr. Top. Membr.* 84, 17–41. doi:10.1016/bs.ctm.2019.07.004

Boucher, E., and Mandato, C. A. (2015). Plasma membrane and cytoskeleton dynamics during single-cell wound healing. *Biochimica Biophysica Acta - Mol. Cell Res.* 1853 (10), 2649–2661. doi:10.1016/j.bbamcr.2015.07.012 Bouter, A., Gounou, C., Bérat, R., Tan, S., Gallois, B., Granier, T., et al. (2011). Annexin-A5 assembled into two-dimensional arrays promotes cell membrane repair. *Nat. Commun.* 2 (1), 270-279. doi:10.1038/ncomms1270

Boye, T. L., and Nylandsted, J. (2016). Annexins in plasma membrane repair. *Biol. Chem.* 397 (10), 961–969. doi:10.1515/hsz-2016-0171

Boyer, L. A., Plath, K., Zeitlinger, J., Brambrink, T., Medeiros, L. A., Lee, T. I., et al. (2006). Polycomb complexes repress developmental regulators in murine embryonic stem cells. *Nature* 441 (7091), 349–353. doi:10.1038/nature04733

Burnstock, G. (2007). Purine and pyrimidine receptors. Cell. Mol. life Sci. 64 (12), 1471–1483. doi:10.1007/s00018-007-6497-0

Cabukusta, B., and Neefjes, J. (2018). Mechanisms of lysosomal positioning and movement. *Traffic* 19 (10), 761–769. doi:10.1111/tra.12587

Cai, C., Masumiya, H., Weisleder, N., Matsuda, N., Nishi, M., Hwang, M., et al. (2009a). MG53 nucleates assembly of cell membrane repair machinery. *Nat. Cell Biol.* 11 (1), 56–64. doi:10.1038/ncb1812

Cai, C., Weisleder, N., Ko, J. K., Komazaki, S., Sunada, Y., Nishi, M., et al. (2009b). Membrane repair defects in muscular dystrophy are linked to altered interaction between MG53, caveolin-3, and dysferlin. *J. Biol. Chem.* 284 (23), 15894–15902. doi:10.1074/jbc.M109.009589

Camussi, G., Deregibus, M. C., Bruno, S., Cantaluppi, V., and Biancone, L. (2010). Exosomes/microvesicles as a mechanism of cell-to-cell communication. *Kidney Int.* 78 (9), 838–848. doi:10.1038/ki.2010.278

Carmeille, R., Bouvet, F., Tan, S., Croissant, C., Gounou, C., Mamchaoui, K., et al. (2016). Membrane repair of human skeletal muscle cells requires Annexin-A5. *Biochimica Biophysica Acta (BBA) - Mol. Cell Res.* 1863 (9), 2267–2279. doi:10.1016/j.bbamcr.2016.06.003

Cenni, V., Sirri, A., Riccio, M., Lattanzi, G., Santi, S., De Pol, A., et al. (2003). Targeting of the Akt/PKB kinase to the actin skeleton. *Cell. Mol. Life Sci.* 60 (12), 2710–2720. doi:10.1007/s00018-003-3349-4

Chang, D. T. W., and Reynolds, I. J. (2006). Differences in mitochondrial movement and morphology in young and mature primary cortical neurons in culture. *Neuroscience* 141 (2), 727–736. doi:10.1016/j.neuroscience.2006.01.034

Chasserot-Golaz, S., Vitale, N., Umbrecht-Jenck, E., Knight, D., Gerke, V., and Bader, M. F. (2005). Annexin 2 promotes the formation of lipid microdomains required for calcium-regulated exocytosis of dense-core vesicles. *Mol. Biol. Cell* 16 (3), 1108–1119. doi:10.1091/mbc.e04-07-0627

Cheng, X., Zhang, X., Yu, L., and Xu, H. (2015). Calcium signaling in membrane repair. Seminars Cell Dev. Biol. 45, 24-31. doi:10.1016/j.semcdb.2015.10.031

Chuo, S. T. Y., Chien, J. C. Y., and Lai, C. P. K. (2018). Imaging extracellular vesicles: Current and emerging methods. *J. Biomed. Sci.* 25 (1), 91–10. doi:10.1186/s12929-018-0494-5

Ciani, E., Guidi, S., Bartesaghi, R., and Contestabile, A. (2002). Nitric oxide regulates cGMP-dependent cAMP-responsive element binding protein phosphorylation and Bcl-2 expression in cerebellar neurons: Implication for a survival role of nitric oxide. *J. Neurochem.* 82 (5), 1282–1289. doi:10.1046/j.1471-4159.2002.01080.x

Clark, A. G., Miller, A. L., Vaughan, E., Yu, H. Y. E., Penkert, R., and Bement, W. M. (2009). Integration of single and multicellular wound responses. *Curr. Biol.* 19 (16), 1389–1395. doi:10.1016/j.cub.2009.06.044

Clark, J. D., Lin, L. L., Kriz, R. W., Ramesha, C. S., Sultzman, L. A., Lin, A. Y., et al. (1991). A novel arachidonic acid-selective cytosolic PLA2 contains a Ca(2+)-dependent translocation domain with homology to PKC and GAP. *Cell* 65 (6), 1043–1051. doi:10. 1016/0092-8674(91)90556-e

Cocucci, E., Racchetti, G., Rupnik, M., and Meldolesi, J. (2008). The regulated exocytosis of enlargeosomes is mediated by a SNARE machinery that includes VAMP4. J. Cell Sci. 121 (18), 2983–2991. doi:10.1242/jcs.032029

Corrotte, M., Almeida, P. E., Tam, C., Castro-Gomes, T., Fernandes, M. C., Millis, B. A., et al. (2013). Caveolae internalization repairs wounded cells and muscle fibers. *eLife* 2013 (2), e00926. doi:10.7554/eLife.00926

Coumans, J. V., Lin, T. T. S., Hai Ning, D., MacArthur, L., McAtee, M., Nash, C., et al. (2001). Axonal regeneration and functional recovery after complete spinal cord transection in rats by delayed treatment with transplants and neurotrophins. *J. Neurosci.* 21 (23), 9334–9344. doi:10.1523/JNEUROSCI.21-23-09334.2001

Darenfed, H., and Mandato, C. A. (2011). Wound-induced contractile ring: A model for cytokinesis. *Biochem. Cell Biol.* 83 (6), 711–720. doi:10.1139/o05-164

Davenport, N. R., Sonnemann, K. J., Eliceiri, K. W., and Bement, W. M. (2016). Membrane dynamics during cellular wound repair. *Mol. Biol. Cell* 27 (14), 2272–2285. doi:10.1091/mbc.E16-04-0223

Davies, K. E., and Nowak, K. J. (2006). Molecular mechanisms of muscular dystrophies: Old and new players. *Nat. Rev. Mol. Cell Biol.* 7 (10), 762–773. doi:10. 1038/nrm2024

De Rooij, J., Zwartkruis, F. J. T., Verheijen, M. H. G., Cool, R. H., Nijman, S. M. B., Wittinghofer, A., et al. (1998). Epac is a Rap1 guanine-nucleotide-exchange factor directly activated by cyclic AMP. *Nature* 396 (6710), 474–477. doi:10.1038/24884 Demonbreun, A. R., Posey, A. D., Heretis, K., Swaggart, K. A., Earley, J. U., Pytel, P., et al. (2010). Myoferlin is required for insulin-like growth factor response and muscle growth. *FASEB J.* 24 (4), 1284–1295. doi:10.1096/fj.09-136309

Demonbreun, A. R., Quattrocelli, M., Barefield, D. Y., Allen, M. V., Swanson, K. E., and McNally, E. M. (2016). An actin-dependent annexin complex mediates plasma membrane repair in muscle. *J. Exp. Med.* 213 (7), 705–718. doi:10.1083/jcb.201512022

Detrait, E. R., Yoo, S., Eddleman, C. S., Fukuda, M., Bittner, G. D., and Fishman, H. M. (2000). Plasmalemmal repair of severed neurites of PC12 cells requires Ca^{2+} and synaptotagmin. *J. Neurosci. Res.* 62, 566–573.

Dias, C., and Nylandsted, J. (2021). Plasma membrane integrity in health and disease: Significance and therapeutic potential. *Cell Discov.* 7 (1), 4–18. doi:10.1038/s41421-020-00233-2

Doherty, K. R., Demonbreun, A. R., Wallace, G. Q., Cave, A., Posey, A. D., Heretis, K., et al. (2008). The endocytic recycling protein EHD2 interacts with myoferlin to regulate myoblast fusion. *J. Biol. Chem.* 283 (29), 20252–20260. doi:10.1074/jbc.M802306200

Draeger, A., Monastyrskaya, K., and Babiychuk, E. B. (2011). Plasma membrane repair and cellular damage control: The annexin survival kit. *Biochem. Pharmacol.* 81 (6), 703–712. doi:10.1016/j.bcp.2010.12.027

Drücker, P., Bachler, S., Wolfmeier, H., Schoenauer, R., Köffel, R., Babiychuk, V. S., et al. (2018). Pneumolysin-damaged cells benefit from non-homogeneous toxin binding to cholesterol-rich membrane domains. *Biochimica Biophysica Acta (BBA) - Mol. Cell Biol. Lipids* 1863 (8), 795–805. doi:10.1016/j.bbalip.2018.04.010

Ebstrup, M. L., Dias, C., Heitmann, A. S. B., Sønder, S. L., and Nylandsted, J. (2021). Actin cytoskeletal dynamics in single-cell wound repair. *Int. J. Mol. Sci.* 22 (19), 10886. doi:10.3390/ijms221910886

Eddleman, C. S., Ballinger, M. L., Smyers, M. E., Fishman, H. M., and Bittner, G. D. (1998a). Endocytotic Formation of vesicles and other membranous structures induced by Ca2+ and axolemmal injury. *J. Neurosci.* 18 (11), 4029–4041. doi:10.1523/ JNEUROSCI.18-11-04029.1998

Eddleman, C. S., Ballinger, M. L., Smyers, M. E., Godell, C. M., Fishman, H. M., and Bittner, G. D. (1997). Repair of plasmalemmal lesions by vesicles. *Proc. Natl. Acad. Sci.* U. S. A. 94 (9), 4745–4750. doi:10.1073/pnas.94.9.4745

Eddleman, C. S., Bittner, G. D., and Fishman, H. M. (2000). Barrier permeability at cut axonal ends progressively decreases until an ionic seal is formed. *Biophysical J.* 79 (4), 1883–1890. doi:10.1016/S0006-3495(00)76438-1

Eddleman, C. S., Bittner, G. D., and Fishman, H. M. (2002). SEM comparison of severed ends of giant axons isolated from squid (Loligo pealeii) and crayfish (*Procambarus clarkii*). *Biol. Bull.* 203 (2), 219–220. doi:10.2307/1543406

Eddleman, C. S., Godell, C. M., Fishman, H. M., Tytell, M., and Bittner, G. D. (1995). Fluorescent labeling of the glial sheath of giant nerve fibers. *Biol. Bull.* 189 (2), 218–219. doi:10.1086/BBLv189n2p218

Eddleman, C. S., Smyers, M. E., Lore, A., Fishman, H. M., and Bittner, G. D. (1998b). Anomalies associated with dye exclusion as a measure of axolemmal repair in invertebrate axons. *Neurosci. Lett.* 256 (3), 123–126. doi:10.1016/s0304-3940(98)00709-5

Fein, A., and Terasaki, M. (2005). Rapid increase in plasma membrane chloride permeability during wound resealing in starfish oocytes. *J. General Physiology* 126 (2), 151–159. doi:10.1085/jgp.200509294

Fishman, H. M., and Bittner, G. D. (2003). Vesicle-mediated restoration of a plasmalemmal barrier in severed axons. *Union Physiol. sci./am. Physiol. Soc. News Physiol. Sci.* 18, 115–118. doi:10.1152/nips.01429.2002

Fishman, H. M., Krause, T. L., Miller, A. L., and Bittner, G. D. (1995). Retardation of the spread of extracellular Ca2+ into transected, unsealed squid giant axons. *Biol. Bull.* 189 (2), 208–209. doi:10.1086/BBLv189n2p208

Fishman, H. M., and Metuzals, J. (1993). Ca2+-Induced axosome formation in internally dialyzed giant axons of Loligo pealei. *Biol. Bull.* 185 (2), 292–293. doi:10. 1086/BBLv185n2p292

Fishman, H. M., Tewari, K. P., and Stein, P. G. (1990). Injury-induced vesiculation and membrane redistribution in squid giant axon. *BBA - Biomembr.* 1023 (3), 421–435. doi:10.1016/0005-2736(90)90135-b

Gabel, M., and Chasserot-Golaz, S. (2016). Annexin A2, an essential partner of the exocytotic process in chromaffin cells. *J. Neurochem.* 137 (6), 890–896. doi:10.1111/jnc. 13628

Gallant, P. E. (1988). Effects of the external ions and metabolic poisoning on the constriction of the squid giant axon after axotomy. *J. Neurosci.* 8 (5), 1479–1484. doi:10. 1523/JNEUROSCI.08-05-01479.1988

Gallant, P. E., and Galbraith, J. A. (1997). Axonal structure and function after axolemmal leakage in the squid giant axon. *J. neurotrauma* 14 (11), 811–822. doi:10. 1089/neu.1997.14.811

Garcia, E. P., McPherson, P. S., Chilcote, T. J., Takei, K., and De Camilli, P. (1995). rbSec1A and B colocalize with syntaxin 1 and SNAP-25 throughout the axon, but are not in a stable complex with syntaxin. *J. Cell Biol.* 129 (1), 105–120. doi:10.1083/jcb.129.1.105

George, E. B., Glass, J. D., and Griffin, J. W. (1995). Axotomy-induced axonal degeneration is mediated by calcium influx through ion-specific channels. *J. Neurosci.* 75 (10), 6445–6452. doi:10.1523/JNEUROSCI.15-10-06445.1995

Gerke, V., Creutz, C. E., and Moss, S. E. (2005). Annexins: Linking Ca2+ signalling to membrane dynamics. *Nat. Rev. Mol. Cell Biol.* 6 (6), 449-461. doi:10.1038/nrm1661

Ghergherehchi, C. L., Mikesh, M., Sengelaub, D. R., Jackson, D. M., Smith, T., Nguyen, J., et al. (2019). Polyethylene glycol (PEG) and other bioactive solutions with neurorrhaphy for rapid and dramatic repair of peripheral nerve lesions by PEG-fusion. *J. Neurosci. Methods* 314, 1–12. doi:10.1016/j.jneumeth.2018.12.015

Godell, C. M., Smyers, M. E., Eddleman, C. S., Ballinger, M. L., Fishman, H. M., and Bittner, G. D. (1997). Calpain activity promotes the sealing of severed giant axons. *Proc. Natl. Acad. Sci. U. S. A.* 94 (9), 4751–4756. doi:10.1073/pnas.94.9.4751

Golan, D. E., Rando, R. R., Alecio, M. R., and Veatch, W. R. (1984). Lateral mobility of phospholipid and cholesterol in the human erythrocyte membrane: Effects of proteinlipid interactions. *Biochemistry* 23 (2), 332–339. doi:10.1021/bi00297a024

Goñi, F. M., and Alonso, A. (1999). Structure and functional properties of diacylglycerols in membranes. *Prog. Lipid Res.* 38 (1), 1–48. doi:10.1016/s0163-7827(98)00021-6

Goñi, F. M., Montes, L. R., and Alonso, A. (2012). Phospholipases C and sphingomyelinases: Lipids as substrates and modulators of enzyme activity. *Prog. Lipid Res.* 51 (3), 238–266. doi:10.1016/j.plipres.2012.03.002

Gordon, L. M., Looney, F. D., and Curtain, C. C. (1985). Spin probe clustering in human-erythrocyte ghosts. J. Membr. Biol. 84 (1), 81–95. doi:10.1007/BF01871650

Grembowicz, K. P., Sprague, D., and McNeil, P. L. (1999). Temporary disruption of the plasma membrane is required for c-fos expression in response to mechanical stress. *Mol. Biol. Cell* 10 (4), 1247–1257. doi:10.1091/mbc.10.4.1247

Han, J. M., Kim, J. H., Lee, B. D., Lee, S. D., Kim, Y., Jung, Y. W., et al. (2002). Phosphorylation-dependent regulation of phospholipase D2 by protein kinase C delta in rat Pheochromocytoma PC12 cells. *J. Biol. Chem.* 277 (10), 8290–8297. doi:10.1074/ jbc.M108343200

Hatakeyama, H., Kishimoto, T., Nemoto, T., Kasai, H., and Takahashi, N. (2006). Rapid glucose sensing by protein kinase A for insulin exocytosis in mouse pancreatic islets. *J. physiology* 570 (2), 271–282. doi:10.1113/jphysiol.2005. 096560

Hendricks, B. K., and Shi, R. (2014). Mechanisms of neuronal membrane sealing following mechanical trauma. *Neurosci. Bull.* 30 (4), 627–644. doi:10.1007/s12264-013-1446-4

Herrmann, A., Arnold, K., and Pratsch, L. (1985). The effect of osmotic pressure of aqueous PEG solutions on red blood cells. *Biosci. Rep.* 5 (8), 689–696. doi:10.1007/BF01117001

Hodgkin, M. N., Clark, J. M., Rose, S., Saqib, K., and Wakelam, M. J. O. (1999). Characterization of the regulation of phospholipase D activity in the detergent-insoluble fraction of HL60 cells by protein kinase C and small G-proteins. *Biochem. J.* 339 (1), 87–93. doi:10.1042/bj3390087

Holopainen, J. M., Angelova, M. I., and Kinnunen, P. K. J. (2000). Vectorial budding of vesicles by asymmetrical enzymatic formation of ceramide in giant liposomes. *Biophysical J.* 78 (2), 830–838. doi:10.1016/S0006-3495(00)76640-9

Horn, A., and Jaiswal, J. K. (2019). Structural and signaling role of lipids in plasma membrane repair. *Curr. Top. Membr.* 84, 67–98. doi:10.1016/bs.ctm.2019. 07.001

Horn, A., Van Der Meulen, J. H., Defour, A., Hogarth, M., Sreetama, S. C., Reed, A., et al. (2017). Mitochondrial redox signaling enables repair of injured skeletal muscle cells. *Sci. Signal.* 10 (495), eaaj1978. doi:10.1126/scisignal.aaj1978

Huang, Y., de Morrée, A., van Remoortere, A., Bushby, K., Frants, R. R., Dunnen, J. T., et al. (2008). Calpain 3 is a modulator of the dysferlin protein complex in skeletal muscle. *Hum. Mol. Genet.* 17 (12), 1855–1866. doi:10.1093/hmg/ddn081

Huang, Y., Laval, S. H., Remoortere, A., Baudier, J., Benaud, C., Anderson, L. V. B., et al. (2007). AHNAK a novel component of the dysferlin protein complex, redistributes to the cytoplasm with dysferlin during skeletal muscle regeneration. *FASEB J.* 21 (3), 732–742. doi:10.1096/fj.06-6628com

Iacovache, I., Degiacomi, M. T., and van der Goot, F. G. (2012). 5.9 pore-forming toxins. *Compr. Biophys.* 5, 164–188.

Jimenez, A. J., Maiuri, P., Lafaurie-Janvore, J., Divoux, S., Piel, M., and Perez, F. (2014). ESCRT machinery is required for plasma membrane repair. *Science* 343 (6174), 1247136. doi:10.1126/science.1247136

Johnson, G. V. W., Litersky, J. M., and Jope, R. S. (1991). Degradation of microtubuleassociated protein 2 and brain spectrin by calpain: A comparative study. *J. Neurochem.* 56 (5), 1630–1638. doi:10.1111/j.1471-4159.1991.tb02061.x

Johnson, J. L., Erickson, J. W., and Cerione, R. A. (2012). C-terminal di-arginine motif of Cdc42 protein is essential for binding to phosphatidylinositol 4,5-bisphosphatecontaining membranes and inducing cellular transformation. J. Biol. Chem. 287 (8), 5764–5774. doi:10.1074/jbc.M111.336487

Kang, W., Robitaille, M. C., Merrill, M., Teferra, K., Kim, C., and Raphael, M. P. (2020). Mechanisms of cell damage due to mechanical impact: An *in vitro* investigation. *Sci. Rep.* 10 (1), 12009. doi:10.1038/s41598-020-68655-2

Kao, C. C., Wrathall, J. R., and Kyoshimov, K. (1983). "Axonal reaction to transection," in *Spinal cord reconstruction*, Editors C. C. Kao, R. P. Bunge, and P. J. Reier (New York: Raven), 41–57.

Kelly, J. P. (1985). "Reactions of neurons to injury," in *Principles of neural science*. Editors E. R. Kandel and J. H. Schwartz (Amsterdam: Elsevier/North Holland), 979.

Koerdt, S. N., Ashraf, A. P. K., and Gerke, V. (2019). Annexins and plasma membrane repair. *Curr. Top. Membr.* 84, 43–65. doi:10.1016/bs.ctm.2019.07.006

Krasnov, P. A., and Enikolopov, G. (2000). Targeting of synaptotagmin to neurite terminals in neuronally differentiated PC12 cells. *J. Cell Sci.* 113 (8), 1389–1404. doi:10. 1242/jcs.113.8.1389

Krause, T. L., and Bittner, G. D. (1990). Rapid morphological fusion of severed myelinated axons by polyethylene glycol. *Proc. Natl. Acad. Sci. U. S. A.* 87 (4), 1471–1475. doi:10.1073/pnas.87.4.1471

Krause, T. L., Fishman, H. M., and Bittner, G. D. (1994). Axolemmal and septal conduction in the impedance of the earthworm medial giant nerve fiber. *Biophysical J.* 67 (2), 692–695. doi:10.1016/S0006-3495(94)80528-4

Krause, T. L., Magarshak, Y., Fishman, H. M., and Bittner, G. D. (1995). Membrane potential and input resistance are ambiguous measures of sealing of transected cable-like structures. *Biophysical J.* 68 (3), 795–799. doi:10.1016/S0006-3495(95)80255-9

Kwik, J., Boyle, S., Fooksman, D., Margolis, L., Sheetz, M. P., and Edidin, M. (2003). Membrane cholesterol, lateral mobility, and the phosphatidylinositol 4,5-bisphosphatedependent organization of cell actin. *Proc. Natl. Acad. Sci.* 100 (24), 13964–13969. doi:10.1073/pnas.2336102100

Lamb, R. G., Harper, C. C., McKinney, J. S., Rzigalinski, B. A., and Ellis, E. F. (1997). Alterations in phosphatidylcholine metabolism of stretch-injured cultured rat astrocytes. J. Neurochem. 68 (5), 1904–1910. doi:10.1046/j.1471-4159.1997.68051904.x

LaPlaca, M. (2009). Plasma membrane damage as a marker of neuronal injury. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. 2009, 1113–1116. doi:10.1109/IEMBS.2009.5334457

Laux, I., Khoshnan, A., Tindell, C., Bae, D., Zhu, X., June, C. H., et al. (2000). Response differences between human CD4(+) and CD8(+) T-cells during CD28 costimulation: Implications for immune cell-based therapies and studies related to the expansion of double-positive T-cells during aging. *Clin. Immunol.* 96 (3), 187–197. doi:10.1006/clim. 2000.4902

Lek, A., Evesson, F. J., Lemckert, F. A., Redpath, G. M. I., Lueders, A. K., Turnbull, L., et al. (2013). Calpains, cleaved mini-DysferlinC72, and L-type channels underpin calcium-dependent muscle membrane repair. *J. Neurosci.* 33 (12), 5085–5094. doi:10.1523/JNEUROSCI.3560-12.2013

Leonard, B. E. (1993). Principles of neural science, third edition. By E. R. Kandel, J. H. Schwartz and T. M. Jessell. Appleton & Lange, 1991. pp. 1135 + xxxvii. ISBN 0-8385-8068-8. *Hum. Psychopharmacol. Clin. Exp.* 8 (4), 294.

Leslie, C. C., and Channon, J. Y. (1990). Anionic phospholipids stimulate an arachidonoyl-hydrolyzing phospholipase A2 from macrophages and reduce the calcium requirement for activity. *Biochimica Biophysica Acta (BBA)/Lipids Lipid Metabolism* 1045 (3), 261–270. doi:10.1016/0005-2760(90)90129-1

Li, P., Bademosi, A. T., Luo, J., and Meunier, F. A. (2018). Actin remodeling in regulated exocytosis: Toward a mesoscopic view. *Trends Cell Biol.* 28 (9), 685–697. doi:10.1016/j.tcb.2018.04.004

Lichstein, J. W., Ballinger, M. L., Blanchette, A. R., Fishman, H. M., and Bittner, G. D. (2000). Structural changes at cut ends of earthworm giant axons in the interval between dye barrier formation and neuritic outgrowth. *J. Comp. Neurology* 416 (2), 143–157. doi:10.1002/(sici)1096-9861(20000110)416:2<143::aid-cne2>3.0.co;2-3

Liu, W. S., and Heckman, C. A. (1998). The sevenfold way of PKC regulation. Cell. Signal. 10 (8), 529-542. doi:10.1016/s0898-6568(98)00012-6

Lopez, S., Bittner, G. D., and Treviño, R. C. (2023). Rapid and effective fusion repair of severed digital nerves using neurorrhaphy and bioengineered solutions including polyethylene glycol: A case report. *Front. Cell. Neurosci.* 16. doi:10.3389/fncel.2022.1087961

Lore, A. B., Hubbell, J. A., Bobb, D. S., Ballinger, M. L., Loftin, K. L., Smith, J. W., et al. (1999). Rapid induction of functional and morphological continuity between severed ends of mammalian or earthworm myelinated axons. *J. Neurosci.* 19 (7), 2442–2454. doi:10.1523/JNEUROSCI.19-07-02442.1999

Mahendra, A., James, H. P., and Jadhav, S. (2019). PEG-grafted phospholipids in vesicles: Effect of PEG chain length and concentration on mechanical properties. *Chem. Phys. Lipids* 218, 47–56. doi:10.1016/j.chemphyslip.2018.12.001

Malinin, V. S., Frederik, P., and Lentz, B. R. (2002). Osmotic and curvature stress affect PEG-induced fusion of lipid vesicles but not mixing of their lipids. *Biophysical J.* 4 (82), 2090–2100. doi:10.1016/S0006-3495(02)75556-2

Mandato, C. A., and Bement, W. M. (2001). Contraction and polymerization cooperate to assemble and close actomyosin rings around Xenopus oocyte wounds. *J. Cell Biol.* 154 (4), 785–797. doi:10.1083/jcb.200103105

Martinez, I., Chakrabarti, S., Hellevik, T., Morehead, J., Fowler, K., and Andrews, N. W. (2000). Synaptotagmin VII regulates Ca2+-dependent exocytosis of lysosomes in fibroblasts. J. Cell Biol. 148 (6), 1141–1149. doi:10.1083/jcb.148.6.1141

McDade, J. R., Archambeau, A., and Michele, D. E. (2014). Rapid actincytoskeleton-dependent recruitment of plasma membrane-derived dysferlin at wounds is critical for muscle membrane repair. *FASEB J.* 28 (8), 3660–3670. doi:10.1096/fj.14-250191 McGill, C. H., Bhupanapadu Sunkesula, S. R., Poon, A. D., Mikesh, M., and Bittner, G. D. (2016). Sealing frequency of B104 cells declines exponentially with decreasing transection distance from the axon hillock. *Exp. Neurol.* 279, 149–158. doi:10.1016/j. expneurol.2016.02.001

McNeil, P. L., and Baker, M. M. (2001). Cell surface events during resealing visualized by scanning-electron microscopy. *Cell Tissue Res.* 304 (1), 141–146. doi:10.1007/s004410000286

McNeil, P. L., and Steinhardt, R. A. (2003). Plasma membrane disruption: Repair, prevention, adaptation. *Annu. Rev. Cell Dev. Biol.* 19 (1), 697–731. doi:10.1146/annurev. cellbio.19.111301.140101

McNeil, P. L., and Terasaki, M. (2001). Coping with the inevitable: How cells repair a torn surface membrane. *Nat. Cell Biol.* 3 (5), E124–E129. doi:10.1038/35074652

Meiri, H., Spira, M. E., and Parnas, I. (1981). Membrane conductance and action potential of a regenerating axonal tip. *Science* 211 (4483), 709–712. doi:10.1126/science. 7455707

Mellgren, R. L., Zhang, W., Miyake, K., and McNeil, P. L. (2007). Calpain is required for the rapid, calcium-dependent repair of wounded plasma membrane. *J. Biol. Chem.* 282 (4), 2567–2575. doi:10.1074/jbc.M604560200

Michailowsky, V., Li, H., Mittra, B., Iyer, S. R., Mazála, D. A. G., Corrotte, M., et al. (2019). Defects in sarcolemma repair and skeletal muscle function after injury in a mouse model of Niemann-Pick type A/B disease. *Skelet. Muscle* 9 (1), 1–15. doi:10.1186/s13395-018-0187-5

Miklavc, P., and Frick, M. (2020). Actin and myosin in non-neuronal exocytosis. *Cells* 9 (6), 1455. doi:10.3390/cells9061455

Millan-Oropeza, A., Rebois, R., David, M., Moussa, F., Dazzi, A., Bleton, J., et al. (2017). Attenuated total reflection fourier transform infrared (ATR FT-IR) for rapid determination of microbial cell lipid content: Correlation with gas chromatographymass spectrometry (GC-MS). *Appl. Spectrosc.* 71 (10), 2344–2352. doi:10.1177/ 0003702817709459

Miyake, K., and McNeil, P. L. (1995). Vesicle accumulation and exocytosis at sites of plasma membrane disruption. *J. Cell Biol.* 131 (6), 1737–1745. doi:10.1083/jcb.131.6. 1737

Mochida, S., Orita, S., Sakaguchi, G., Sasaki, T., and Takai, Y. (1998). Role of the Doc2 alpha-Munc13-1 interaction in the neurotransmitter release process. *Proc. Natl. Acad. Sci. U. S. A.* 95 (19), 11418–11422. doi:10.1073/pnas.95.19.11418

Moe, A., Golding, A. E., and Bement, W. M. (2015). Cell healing: Calcium, repair and regeneration. Seminars Cell & Dev. Biol. 45, 18–23. doi:10.1016/j.semcdb.2015.09.026

Monastyrskaya, K., Babiychuk, E. B., Hostettler, A., Rescher, U., and Draeger, A. (2007). Annexins as intracellular calcium sensors. *Cell Calcium* 41 (3), 207–219. doi:10. 1016/j.ceca.2006.06.008

Money, N. P. (1989). Osmotic pressure of aqueous polyethylene glycols' relationship between molecular weight and vapor pressure deficit. *Plant Physiol.* 91, 766–769. doi:10. 1104/pp.91.2.766

Neary, J. T., Kang, Y., Tran, M., and Feld, J. (2005). Traumatic injury activates protein kinase B/Akt in cultured astrocytes: Role of extracellular ATP and P2 purinergic receptors. *J. Neurotrauma* 22 (4), 491–500. doi:10.1089/neu.2005. 22.491

Nguyen, M. P., Bittner, G. D., and Fishman, H. M. (2005). Critical interval of somal calcium transient after neurite transection determines B104 cell survival. *J. Neurosci. Res.* 81, 805. doi:10.1002/jnr.20606

Nishimura, Y., and Lemasters, J. J. (2001). Glycine blocks opening of a death channel in cultured hepatic sinusoidal endothelial cells during chemical hypoxia. *Cell Death Differ.* 8 (8), 850–858. doi:10.1038/sj.cdd.4400877

Obis, T., Besalduch, N., Hurtado, E., Nadal, L., Santafe, M. M., Garcia, N., et al. (2015). The novel protein kinase C epsilon isoform at the adult neuromuscular synapse: Location, regulation by synaptic activity-dependent muscle contraction through TrkB signaling and coupling to ACh release. *Mol. Brain* 8 (1), 8–16. doi:10.1186/ s13041-015-0098-x

Papalazarou, V., and Machesky, L. M. (2021). The cell pushes back: The Arp2/ 3 complex is a key orchestrator of cellular responses to environmental forces. *Curr. Opin. Cell Biol.* 68, 37–44. doi:10.1016/j.ceb.2020.08.012

Poon, A. D., McGill, S. H., Raju, S., Sunkesula, B., Burgess, Z. S., Dunne, P. J., et al. (2018). Ca2+/calmodulin-dependent protein kinase II and Dimethyl Sulfoxide affect the sealing frequencies of transected hippocampal neurons. *J. Neurosci. Res.* 96 (7), 1208–1222. doi:10.1002/jnr.24232

Post, S. R., Christian Rump, L., Zambon, A., Hughes, R. J., Buda, M. D., Paul Jacobson, J., et al. (1998). ATP Activates cAMP Production via Multiple Purinergic Receptors in MDCK-D1 Epithelial Cells: Blockade of an autocrine/paracrine pathway to define receptor preference of an agonist. *J. Biol. Chem.* 273 (36), 23093–23097. doi:10.1074/jbc. 273.36.23093

Purves, D., Augustine, G., Hall, W., LaMantia, A., Mooney, R., Platt, M., et al. (2018). *Neuroscience.* 6th edn. New York: Sinauer Associates.

Ramón Y Cajal, S., DeFelipe, J., Jones, E. G., and May, R. M. (1991). "Cajal's degeneration and regeneration of the nervous system," in *Cajal's degeneration and*

regeneration of the nervous system (Oxford, UK: Oxford University Press), 1–786. doi:10.1093/acprof:oso/9780195065169.001.0001

Raucher, D., Stauffer, T., Chen, W., Shen, K., Guo, S., York, J. D., et al. (2000). Phosphatidylinositol 4,5-bisphosphate functions as a second messenger that regulates cytoskeleton-plasma membrane adhesion. *Cell* 100 (2), 221–228. doi:10.1016/s0092-8674(00)81560-3

Reddy, A., Caler, E. V., and Andrews, N. W. (2001). Plasma membrane repair is mediated by Ca2+-regulated exocytosis of lysosomes. *Cell* 106 (2), 157–169. doi:10. 1016/s0092-8674(01)00421-4

Rintoul, G. L., and Reynolds, I. J. (2010). Mitochondrial trafficking and morphology in neuronal injury. *Biochimica Biophysica Acta (BBA) - Mol. Basis Dis.* 1802 (1), 143–150. doi:10.1016/j.bbadis.2009.09.005

Riske, K. A., and Dimova, R. (2005). Electro-deformation and poration of giant vesicles viewed with high temporal resolution. *Biophysical J.* 88 (2), 1143–1155. doi:10. 1529/biophysj.104.050310

Rohatgi, R., Ma, L., Miki, H., Lopez, M., Kirchhausen, T., Takenawa, T., et al. (1999). The interaction between N-WASP and the Arp2/3 complex links Cdc42dependent signals to actin assembly. *Cell* 97 (2), 221–231. doi:10.1016/s0092-8674(00)80732-1

Rosengarth, A., Gerke, V., and Luecke, H. (2001). X-ray structure of full-length annexin 1 and implications for membrane aggregation. J. Mol. Biol. 306 (3), 489–498. doi:10.1006/jmbi.2000.4423

Roth, M. G., Bi, K., Ktistakis, N. T., and Yu, S. (1999). Phospholipase D as an effector for ADP-ribosylation factor in the regulation of vesicular traffic. *Chem. Phys. Lipids* 98 (1-2), 141–152. doi:10.1016/s0009-3084(99)00026-2

Sarasa-Renedo, A., Tunç-Civelek, V., and Chiquet, M. (2006). Role of RhoA/ROCKdependent actin contractility in the induction of tenascin-C by cyclic tensile strain. *Exp. Cell Res.* 312 (8), 1361–1370. doi:10.1016/j.yexcr.2005.12.025

Sassone-Corsi, P., Visvader, J., Ferland, L., Mellon, P. L., and Verma, I. M. (1988). Induction of proto-oncogene fos transcription through the adenylate cyclase pathway: Characterization of a cAMP-responsive element. *Genes & Dev.* 2 (12A), 1529–1538. doi:10.1101/gad.2.12a.1529

Sattler, R., Tymianski, M., Feyaz, I., Hafner, M., and Tator, C. H. (1996). Voltagesensitive calcium channels mediate calcium entry into cultured mammalian sympathetic neurons following neurite transection. *Brain Res.* 719 (1-2), 239–246. doi:10.1016/0006-8993(96)00125-4

Scheffer, L. L., Sreetama, S. C. H., Sharma, N., Medikayala, S., Brown, K. J., Defour, A., et al. (2014). Mechanism of Ca²⁺-triggered ESCRT assembly and regulation of cell membrane repair. *Nat. Commun.* 5, 5646. doi:10.1038/ncomms6646

Schlaepfer, W. W., and Bunge, R. P. (1973). Effects of calcium ion concentration on the degeneration of amputated axons in tissue culture. *J. Cell Biol.* 59 (2), 456–470. doi:10.1083/jcb.59.2.456

Schöneberg, J., Lee, I. H., Iwasa, J. H., and Hurley, J. H. (2016). Reverse-topology membrane scission by the ESCRT proteins. *Nat. Publ. Group* 18, 5. doi:10.1038/nrm. 2016.121

Schuh, M. (2011). An actin-dependent mechanism for long-range vesicle transport. Nat. Cell Biol. 13 (12), 1431–1436. doi:10.1038/ncb2353

Siman, R., Noszek, J. C., and Kegerise, C. (1989). Calpain I activation is specifically related to excitatory amino acid induction of hippocampal damage. *J. Neurosci.* 9 (5), 1579–1590. doi:10.1523/JNEUROSCI.09-05-01579.1989

Singer, S. J., and Nicolson, G. L. (1972). The fluid mosaic model of the structure of cell membranes. *Science* 175 (4023), 720–731. doi:10.1126/science.175. 4023.720

Sokac, A. M., Co, C., Taunton, J., and Bement, W. (2003). Cdc42-dependent actin polymerization during compensatory endocytosis in Xenopus eggs. *Nat. Cell Biol.* 5 (8), 727–732. doi:10.1038/ncb1025

Sønder, S. L., Häger, S. C., Heitmann, A. S. B., Frankel, L. B., Dias, C., Simonsen, A. C., et al. (2021). Restructuring of the plasma membrane upon damage by LC3-associated macropinocytosis. *Sci. Adv.* 7 (27), eabg1969. doi:10.1126/sciadv.abg1969

Song, G., Ouyang, G., and Bao, S. (2005). The activation of Akt/PKB signaling pathway and cell survival. *J. Cell. Mol. Med.* 9 (1), 59–71. doi:10.1111/j.1582-4934.2005. tb00337.x

Spaeth, C. S., Boydston, E. A., Figard, L. R., Zuzek, A., and Bittner, G. D. (2010). A model for sealing plasmalemmal damage in neurons and other eukaryotic cells. *J. Neurosci.* 30 (47), 15790–15800. doi:10.1523/JNEUROSCI.4155-10.2010

Spaeth, C. S., Fan, J. D., Spaeth, E. B., Robison, T., Wilcott, R. W., and Bittner, G. D. (2012a). Neurite transection produces cytosolic oxidation, which enhances plasmalemmal repair. *J. Neurosci. Res.* 90 (5), 945–954. doi:10.1002/jnr.22823

Spaeth, C. S., Robison, T., Fan, J. D., and Bittner, G. D. (2012b). Cellular mechanisms of plasmalemmal sealing and axonal repair by polyethylene glycol and methylene blue. *J. Neurosci. Res.* 90 (5), 955–966. doi:10.1002/jnr.23022

Spaeth, C. S., Spaeth, E. B., Wilcott, R. W., Fan, J. D., Robison, T., and Bittner, G. D. (2012c). Pathways for plasmalemmal repair mediated by PKA, Epac, and cytosolic

oxidation in rat B104 cells *in vitro* and rat sciatic axons *ex vivo*. *Dev. Neurobiol*. 72 (11), 1399–1414. doi:10.1002/dneu.20998

Spira, M. E., Benbassat, D., and Dormann, A. (1993). Resealing of the proximal and distal cut ends of transected axons: Electrophysiological and ultrastructural analysis. *J. Neurobiol.* 24 (3), 300–316. doi:10.1002/neu.480240304

Spötl, L., Sarti, A., Dierich, M. P., and Möst, J. (1995). Cell membrane labeling with fluorescent dyes for the demonstration of cytokine-induced fusion between monocytes and tumor cells. *Cytometry* 21 (2), 160–169. doi:10.1002/cyto.990210208

Sridhar, K. C., Hersch, N., Dreissen, G., Merkel, R., and Hoffmann, B. (2020). Calcium mediated functional interplay between myocardial cells upon laser-induced single-cell injury: An *in vitro* study of cardiac cell death signaling mechanisms. *Cell Commun. Signal.* 18 (1), 191. doi:10.1186/s12964-020-00689-5

Stassen, M., Müller, C., Richter, C., Neudörfl, C., Hültner, L., Bhakdi, S., et al. (2003). The streptococcal exotoxin streptolysin O activates mast cells to produce tumor necrosis factor alpha by p38 mitogen-activated protein kinase- and protein kinase C-dependent pathways. *Infect. Immun.* 71 (11), 6171–6177. doi:10.1128/IAI.71.11.6171-6177.2003

Steinhardt, R. A., Bi, G., and Alderton, J. M. (1994). Cell membrane resealing by a vesicular mechanism similar to neurotransmitter release. *Science* 263 (5145), 390–393. doi:10.1126/science.7904084

Ster, J., De Bock, F., Bertaso, F., Abitbol, K., Daniel, H., Bockaert, J., et al. (2009). Epac mediates PACAP-dependent long-term depression in the hippocampus. *J. Physiology* 587 (1), 101–113. doi:10.1113/jphysiol.2008.157461

Strautman, A. F., Cork, R. J., and Robinson, K. R. (1990). The distribution of free calcium in transected spinal axons and its modulation by applied electrical fields. *J. Neurosci.* 10 (11), 3564–3575. doi:10.1523/JNEUROSCI.10-11-03564.1990

Tam, C., Idone, V., Devlin, C., Fernandes, M. C., Flannery, A., He, X., et al. (2010). Exocytosis of acid sphingomyelinase by wounded cells promotes endocytosis and plasma membrane repair. J. Cell Biol. 189 (6), 1027–1038. doi:10.1083/jcb.201003053

Tan, Y., Rouse, J., Zhang, A., Cariati, S., Cohen, P., and Comb, M. J. (1996). FGF and stress regulate CREB and ATF-1 via a pathway involving p38 MAP kinase and MAPKAP kinase-2. *EMBO J.* 15 (17), 4629–4642. doi:10.1002/j.1460-2075.1996.tb00840.x

Terasaki, M., Miyake, K., and McNeil, P. L. (1997). Large plasma membrane disruptions are rapidly resealed by Ca²⁺-dependent vesicle-vesicle fusion events. *J. Cell Biol.* 139 (1), 63–74. doi:10.1083/jcb.139.1.63

Togo, T., Alderton, J. M., Bi, G. Q., and Steinhardt, R. A. (1999). The mechanism of facilitated cell membrane resealing. *J. Cell Sci.* 112 (5), 719–731. doi:10.1242/jcs.112. 5.719

Togo, T., Alderton, J. M., and Steinhardt, R. A. (2003). Long-term potentiation of exocytosis and cell membrane repair in fibroblasts. *Mol. Biol. Cell* 14, 93–106. doi:10. 1091/mbc.e02-01-0056

Togo, T. (2019). Autocrine purinergic signaling stimulated by cell membrane disruption is involved in both cell membrane repair and adaptive response in MDCK cells. *Biochem. Biophysical Res. Commun.* 511 (1), 161–164. doi:10.1016/j. bbrc.2019.02.061

Togo, T. (2017). Cell membrane disruption stimulates cAMP and Ca2+ signaling to potentiate cell membrane resealing in neighboring cells. *Biol. Open* 6 (12), 1814–1819. doi:10.1242/bio.028977

Togo, T. (2012). Cell membrane disruption stimulates NO/PKG signaling and potentiates cell membrane repair in neighboring cells. *PLoS One* 7 (8), e42885. doi:10.1371/journal.pone.0042885

Togo, T. (2006). Disruption of the plasma membrane stimulates rearrangement of microtubules and lipid traffic toward the wound site. *J. Cell Sci.* 119 (13), 2780–2786. doi:10.1242/jcs.03006

Togo, T., Krasieva, T. B., and Steinhardt, R. A. (2000). A decrease in membrane tension precedes successful cell-membrane repair. *Mol. Biol. Cell* 11, 4339–4346. doi:10. 1091/mbc.11.12.4339

Togo, T. (2004). Long-term potentiation of wound-induced exocytosis and plasma membrane repair is dependent on cAMP-response element-mediated transcription via a protein kinase C- and p38 MAPK-dependent pathway. *J. Biol. Chem.* 279 (43), 44996–45003. doi:10.1074/jbc.M406327200

Togo, T. (2014). Short-term potentiation of membrane resealing in neighboring cells is mediated by purinergic signaling. *Purinergic Signal.* 10 (2), 283–290. doi:10.1007/s11302-013-9387-y

Togo, T., and Steinhardt, R. A. (2004). Nonmuscle myosin IIA and IIB have distinct functions in the exocytosis-dependent process of cell membrane repair. *Mol. Biol. Cell* 15, 688–695. doi:10.1091/mbc.e03-06-0430

Torregrosa-Hetland, C. J., Villanueva, J., Giner, D., Lopez-Font, I., Nadal, A., Quesada, I., et al. (2011). The F-actin cortical network is a major factor influencing the organization of the secretory machinery in chromaffin cells. *J. Cell Sci.* 124 (5), 727–734. doi:10.1242/jcs.078600

Trump, B. F., and Berezesky, I. K. (1995). Calcium-mediated cell injury and cell death. *FASEB J.* 9 (2), 219–228. doi:10.1096/fasebj.9.2.7781924

Um, D. E., Shin, H., Park, D., Ahn, J. M., Kim, J., Song, H., et al. (2020). Molecular analysis of lipid uptake-and necroptosis-associated factor expression in vitrified-warmed mouse oocytes. *Reproductive Biol. Endocrinol.* 18 (1), 37–13. doi:10.1186/s12958-020-00588-x

Van Nest, D. S., Kahan, D. M., and Ilyas, A. M. (2021). Polyethylene glycol fusion of nerve injuries: Review of the technique and clinical applicability. *J. Hand Microsurg.* 13 (2), 49–54. doi:10.1055/s-0040-1718651

Vargas, S. A., and Bittner, G. D. (2019). Natural mechanisms and artificial PEGinduced mechanism that repair traumatic damage to the plasmalemma in eukaryotes. *Curr. Top. Membr.* 84, 129–167. doi:10.1016/bs.ctm.2019.07.005

Vaughan, E. M., You, J. S., Yu, H. Y. E., Lasek, A., Vitale, N., Hornberger, T. A., et al. (2014). Lipid domain-dependent regulation of single-cell wound repair. *Mol. Biol. Cell* 25 (12), 1867–1876. doi:10.1091/mbc.E14-03-0839

Walev, I., Bhakdi, S. C., Hofmann, F., Djonder, N., Valeva, A., Aktories, K., et al. (2001). Delivery of proteins into living cells by reversible membrane permeabilization with streptolysin-O. *Proc. Natl. Acad. Sci. U. S. A.* 98 (6), 3185–3190. doi:10.1073/pnas. 051429498

Xiea, X. Y., and Barrett, J. N. (1991). Membrane resealing in cultured rat septal neurons after neurite transection: Evidence for enhancement by Ca(2+)-triggered protease activity and cytoskeletal disassembly. *J. Neurosci.* 11 (10), 3257–3267. doi:10.1523/JNEUROSCI.11-10-03257.1991

Xu, K., Zhong, G., and Zhuang, X. (2013). Actin, spectrin, and associated proteins form a periodic cytoskeletal structure in axons. *Science* 339 (6118), 452–456. doi:10. 1126/science.1232251

Yang, J., and Shen, M. H. (2006). Polyethylene glycol-mediated cell fusion. Methods Mol. Biol. Clift. N.J.) 325, 59–66. doi:10.1385/1-59745-005-7:59

Yawo, H., and Kuno, M. (1985). Calcium dependence of membrane sealing at the cut end of the cockroach giant axon. *J. Neurosci.* 5 (6), 1626–1632. doi:10.1523/ JNEUROSCI.05-06-01626.1985

Yawo, H., and Kuno, M. (1983). How a nerve fiber repairs its cut end: Involvement of phospholipase A2. *Science* 222 (4630), 1351–1353. doi:10.1126/science.6658457

Yoo, S., Bottenstein, J. E., Bittner, G. D., and Fishman, H. M. (2004). Survival of mammalian B104 cells following neurite transection at different locations depends on somal Ca2+ concentration. *J. Neurobiol.* 60 (2), 137–153. doi:10.1002/neu.20005

Yoo, S., Nguyen, M. P., Fukuda, M., Bittner, G. D., and Fishman, H. M. (2003). Plasmalemmal sealing of transected mammalian neurites is a gradual process mediated by Ca2+-regulated proteins. *J. Neurosci. Res.* 74 (4), 541–551. doi:10. 1002/jnr.10771

Yoshihara, M., Suzuki, K., and Kidokoro, Y. (2000). Two independent pathways mediated by cAMP and protein kinase A enhance spontaneous transmitter release atDrosophila neuromuscular junctions. *J. Neurosci.* 20 (22), 8315–8322. doi:10.1523/ INEUROSCI.20-22-08315.2000

Ziv, N. E., and Spira, M. E. (1995). Axotomy induces a transient and localized elevation of the free intracellular calcium concentration to the millimolar range. *J. Neurophysiology* 74 (6), 2625–2637. doi:10.1152/jn.1995.74.6.2625

Zuzek, A., Fan, J. D., Spaeth, C. S., and Bittner, G. D. (2012). Sealing of transected neurites of rat B104 cells requires a diacylglycerol PKC-dependent pathway and a PKA-dependent pathway. *Cell. Mol. Neurobiol.* 3333 (11), 31–46. doi:10.1007/s10571-012-9868-5